Contents lists available at ScienceDirect

# Heliyon



journal homepage: www.cell.com/heliyon

# Research article

# Earnings management in socially responsible firms around seasoned equity offerings: Evidence from France, Germany, Italy and Spain

# Mariano González-Sánchez, Ana I. Segovia San Juan<sup>\*</sup>, Eva M. Ibáñez Jiménez

Department of Business and Accounting, UNED (National Distance Education University), Paseo Senda del Rey, 11, 28040, Madrid, Spain

# ARTICLE INFO

Keywords: Corporate social responsibility Discretionary accruals Earnings management Real earnings manipulation Seasoned equity offerings

# ABSTRACT

Earnings manipulation (EM) has been a matter of interest to researchers for decades. How this is measured or the motivations of managers to engage in such actions have been studied in detail. Some studies find that managers have incentives to manipulate the earnings that accompany financing activities such as seasoned equity offerings (SEO). Under the corporate social responsibility (CSR) approach, profit manipulation actions have been shown to be mitigated in socially responsible companies. To the best of our knowledge, there are no studies that analyse whether CSR mitigate EM actions in a SEO context. Our work contributes to filling this gap. We investigate whether socially responsible companies exhibit EM in periods prior to SEO. This study uses a panel data model of listed non-financial firms from countries with the same currency and similar accounting rules (France, Germany, Italy and Spain) between 2012 and 2020. Our results show that in all the countries analysed, except Spain, there is a manipulation of operating cash flows in the year prior to capital increases, and only in French companies is there a decrease in the management of this variable in companies with higher development of corporate social responsibility.

# 1. Introduction

Earnings management (EM) can be defined as the purposeful intervention in the elaboration financial reporting process, with the intent of obtaining some private gain [1,2] and reducing the quality of financial reporting. EM have been widely analysed in the literature under different approaches, such as managers manipulating real activities to avoid reporting annual losses [3,4] and accrual-based earnings management through changes in estimates and accounting policies [5–7]. Therefore, if directors deliberately alter their investment decisions by manipulating actual activities, or alter estimates or accounting methods to influence accruals, we are in an EM environment. The motivations for these practices can be stock market reasons related to stock price movements or analysts' judgments, contracting motivations - both lending and incentive motivations - or regulatory reasons [8]. EM activities are often examined under an agency approach, as they are usually motivated by conflicts of interest between ownership and control, where asymmetric information exists [9–11].

Some studies show that companies use EM to manipulate share prices or to align reported results with investor expectations [6,

\* Corresponding author.

https://doi.org/10.1016/j.heliyon.2023.e15171

Received 14 July 2022; Received in revised form 27 March 2023; Accepted 28 March 2023

Available online 5 April 2023



*E-mail addresses*: mariano.gonzalez@cee.uned.es (M. González-Sánchez), asegovia@cee.uned.es (A.I. Segovia San Juan), eibanez@cee.uned.es (E.M. Ibáñez Jiménez).

<sup>2405-8440/© 2023</sup> The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/4.0/).

12–17]. EM generally has a negative impact on the quality of financial reporting, which can reduce the value of the company, causing stakeholders to lose confidence. EM has also been analysed regarding other issues such as insider trading [18], mergers and acquisitions [19], debt issuance [20], or bankruptcy risk [21].

In capital markets, there has been a particular interest in analysing EM around capital increases. Authors such as [22–24], showed that the operating performance of issuing companies showed a substantial improvement before the offering, and that this performance declined in subsequent periods. This pattern was analysed by [25–27], among others, who studied the relationship between EM and equity issues, concluding that issuing firms deliberately manipulate earnings prior to equity offerings. Previous studies have documented that issuing companies manipulate earnings prior to SEOs. When an SEO occurs, investors pay close attention to the company and its share price, and managers have inventive ways to inflate its earnings, the overvaluation of which translates into a transfer of wealth to the company's current and future shareholders [28]. SEOs thus become a particularly favourable context for analysing EM. Such actions around SEOs are negative signals and induce investors to be optimistic about future earnings and overvalue these firms [29]. Managerial opportunism is the factor that is often pointed to explain these behaviours that meet both the immediate needs of the firm as well as its long-term financing needs [30]. The literature has demonstrated that firms use different strategies to manage their earnings around SEOs: using accruals [31,32], managing real earnings [33] or a combination of both procedures [34,35].

Corporate Social Responsibility (CSR) has played a relevant role in business strategy for decades. It has been shown that CSR strategy is positively valued by investors, has a positive effect on corporate reputation [36], and that its practices and disclosure help to reduce asymmetric information in the capital market [37,38]. SEOs are corporate events that play a key role for companies in facilitating external financing activities in the stock market [39] and, in SEO context, there is an information asymmetry between the issuer and external investors [40]. Feng et al. [41] study whether good CSR practices help the firm to reduce this information asymmetry in the SEO context. They conclude that issuers with high CSR scores provide more transparent financial reporting and add value to their shareholders by reducing this asymmetry.

Some studies have shown that companies carry out EM actions in an SEO context. Conversely, most of the literature finds that companies committed to CSR do not engage in EM actions, and conclude that managers of socially responsible companies are less likely to use earnings manipulation strategies [42–45]. Given the relevance of SEO in capital markets, and the influence of CSR on investor behaviour, it is of interest to analyse whether companies considered socially responsible perform EM actions in an SEO context. As far as we know, there are not studies that analyse this issue and our research contribute to filling this gap. Therefore, our aim is to investigate whether, in the SEO environment, the behaviour pattern of CSR companies is similar to that of the rest of the companies (non-CSR companies) or, on the contrary, the fact of being considered socially responsible is a determining factor for not engaging in such practices.

Our empirical analysis focuses on panel data consisting of listed non-financial companies from the largest EU countries (France, Germany, Italy and Spain), for 2012–2020 period, since their financial statement are formulated in the same currency (euro) and are subject to European Directives and IFRS. Besides, we use two measures as a proxy of each company's CSR performance: the inclusion in the Dow Jones Sustainability Index Europe and the Global Reporting Initiative level of non-financial reporting.

In our analysis we study both real earnings manipulation and accrual-based earnings manipulation practices (total earnings management-TEM) and observe that for Spanish companies there is no relationship between TEM or any of its components with subsequent years' capital operations (SEO) or with the quality and development of CSR. For the remaining countries (France, Germany and Italy), we only find a positive relationship between subsequent years' capital operations and discretionary operating cash flows, i. e., companies in these countries increase their discretionary operating cash flows two years before SEO. We also observe that only French companies show a statistically significant and negative relationship between equity issuance and EM.

The rest of study is organised as follows: Section 2 reviews the literature and develops the study hypothesis. Section 3 explains the methodology used to test the hypothesis. Section 4 describes the data. Section 5 shows the results obtained from the estimated models. Section 6 presents the main conclusions.

## 2. Literature review and hypothesis development

Schipper [1] points out that EM could occur in any part of the external disclosure process, and could take several forms such as different accrual-based earnings management (AEM) and real earnings management (REM). Both meanings have been widely explored and raise substantial implications. AEM occurs when managers control reported earnings by exploiting accounting discretion [1,6,8, 46]. Thus, AEM is achieved by modifying the accounting methods or estimates used when introducing a given transaction in the financial statements, for example, changing the method of depreciation of fixed assets or the estimate of the impairment of doubtful accounts.

REM is defined by Roychowdhury [4] as "departures from normal operational practices, motivated by desire of managers to mislead at least some stakeholders into believing certain financial reporting goals have been met in the normal course of operations" (p.337), and can appear in many forms, such as offering price discounts to temporarily increase sales, accelerating sales, engaging in overproduction to lower cost of goods sold, and reducing or delaying discretionary expenditures to improve margins. Al-Shattarat et al. [47] explain REM under opposing approaches such as the signalling theory and the opportunistic theory. According to signalling theory, companies use REM to indicate good future performance [3,4,48]. From the opportunistic approach, managers use REM to manipulate results and deliberately mislead external investors, which could lead to a further decline in future performance. Thus, companies that use REM will experience a negative impact on the subsequent value of the company and its performance.

Managers can also use both practices simultaneously. Zang [49] analyzes the trade-offs between accrual manipulations and REM and concludes that high (low) real activities manipulation realized is directly offset by a lower (higher) amount of AEM. In addition,

executive surveys conducted by Graham et al. [50] suggest that managers emphasize real economic actions rather than exercise accounting discretion to hit earnings benchmarks.

Literature has studied the relationship between CSR and performance in business management [51]. While some authors suggest that socially responsible companies tend to build long-term relationships with stakeholders rather than maximise short-term profits, others argue that CSR is a practice used by managers who engage in opportunistic behaviour [52].

Heal [53] and Harjoto and Jo [54] noted that socially responsible companies make better use of resources and innovation, which translates into an increase in the value of the company. Chen et al. [55] found that the influence of CSR on stock performance depends on the corporate value of the company, and show that CSR activities of companies with low corporate value have a negative effect on stock performance. Conversely, companies with high corporate value have a greater ability to implement CSR, and CSR investments can increase their share prices. Lopatta et al. [56] study CSR under the opportunistic theory and disaggregate the CSR data into a normal and an abnormal component that is subject to agency problems. They find that only normal CSR is not negatively associated with financial performance, while the relationship between abnormal CSR and financial performance is negative.

Recently, environmental, social and governance (ESG) factors have been developed as a subset of CSR [57], and we also find several papers linking ESG and firm financial performance. For instance, Friede et al. [58] analyse 2,000 empirical studies published since the 1970s and conclude that approximately 90% of papers find a non-negative relationship between ESG and financial performance, and the vast majority of studies report positive results, with this relationship being stable over time.

With regard to the object of this empirical research, relationship between CSR and earnings management, there is an extensive literature that found empirical evidence. Chih et al. [59] conclude that companies committed to CSR tend not to smooth earnings; in the similar direction, authors such as [42,43,45,60] find that firms engaged in CSR are less profit management, and that CSR moderates the negative impact of EM on financial performance [61]. Rahmawati et al. [44] show that higher level of real manipulation in cash flow from operations has a negative effect on the relationship between CSR and financial performance. Bozzolan et al. [62] conclude that CSR-oriented firms are less likely to participate in real activity manipulation than accrual manipulation, and this orientation acts as a restriction for REM, which contributes to the creation of value for stakeholders. Almahrog et al. [63] reveal that companies with a higher commitment to CSR are less likely to manage earnings through accruals. Moratis and Egmond [64] prove no relationship between EM and CSR, and that firms with high environmental impact do not appear to practice profit management, but show higher CSR performance. Dimitropoulos [65] study the impact of CSR performance on EM behaviour using a sample from 24 European Union (EU) countries over the period 2003–2018, and conclude that there is a negative association between CSR and EM, so that companies with high CSR performance are associated with higher quality financial reporting. Chen and Hung [66] argue that the company engages in CSR not only to increase the transparency of CSR information and to create interaction with stakeholders, but also to reduce the phenomenon of EM. When the firm's CSR performance is lower, it is susceptible to EM. Moreover, CSR performance can increase the value of the firm, but if the firm uses CSR to cover up and distract attention from management manipulation of profits, the value of the firm will decrease. In the same vein, Palacios-Manzano et al. [52] find that firms that are more committed to CSR are less involved in EM. This negative relationship between EM and CSR extends to other studies that consider environmental, social and governance (ESG) information, and reach similar conclusions ([67–70]. Borralho et al. [71] independently considered the influence of ESG disclosure's three components on EM and found that environmental disclosure is positively associated with EM, while social disclosure and governance disclosures are negatively associated with EM. Nagy et al. [72] find that CSR performance of companies is positively correlated with the volume of profits earned and negatively correlated with the level of discretionary accruals. Khenissi et al. [73] relate the inclusion of CSR criteria in compensation contracts, observing a negative relationship with EM. Chouaibi and Zouari [69] analyse a sample of 540 European ESG firms over the period 2011-2019, concluding that the more important CSR practices are, the less the firm engages in an aggressive earnings management strategy, which negatively affects its cost of capital cost of equity. Khlifi and Zourari [74] show that CSR is significantly and negatively associated with REM in the most innovative companies and, the same authors, show that good corporate governance takes on a moderating role in the relationship between CSR and REM [75]. The role of gender policies has also been analysed and authors such as [76] confirm that companies that participate in CSR activities are less motivated to practice EM and are more likely to produce high quality financial reports. At the same time, gender diversity increases board effectiveness by reducing the level of earnings management, ensuring high quality financial reporting and developing CSR activities.

In contrast, Prior et al. [77] reveal a positive association between EM and CSR practices and conclude that CSR companies may be involved in earnings management practices, and that the combination of EM and CSR causes a decrease in the company's financial performance; in the banking sector, Grougiou et al. [78] prove a positive significant relationship between CSR and EM. Martínez-Ferrero et al. [36] demonstrate that CSR activities can be used to hide EM practices. More recently, López-González et al. [79] conclude that companies with greater social responsibility activity show a higher manipulative behaviour through the promotion of actions that mask the company's financial and economic reality, with this positive relationship moderating downwards in family firms.

Regarding the relationship between capital increases and EM, Rangan [25] and Teoh et al. [26], among others, have shown that in the run-up to a capital increase, companies tend to exaggerate their earnings by providing the investor with a misleading view of their situation, and that there is a negative relationship between the returns obtained around the offerings and the profits obtained afterwards, indicating that companies deliberately manipulate earnings prior to share offerings. Cohen and Zarowin [34] show that firms engage in both forms of EM in the years prior to SEO, and find that the decline in operating performance after SEO attributable to actual activity management is more severe than that attributable to accruals management. The reasons for this behavior and how to mitigate it have also been analysed. Kim [17] shows that in firms that raise capital, managers with greater accounting flexibility are more likely to provide more optimistic forecasts than analysts. Yang et al. [29] discover that financial constraints and distress risk are two motives that drive firms to report broader earnings around SEO. From UE companies data, Fauver et al. [80] note that the implementation of

the Market Abuse Directive and the Prospectus Directive improves the transparency and quality of company reporting which helps to reduce incentives for managers to manipulate profits around public equity issues. Chang and Lin [35] show that firms will use real, as well as accrual-based, earnings management activities around SEO. In general, companies planning to issue new shares have more incentives to improve their image. Overly optimistic earnings forecasts, combined with EM, provide a useful tool to manipulate investor expectations and are likely to lead to a decline in earnings after the capital increase [81]. Bertomeu et al. [82] conclude that there is more EM during SEO in smaller and growing firms and in industries with more irregularities. Opare et al. [83] show that post-SEO performance is worse for firms that engage in RAM than for those that manipulate accruals, that the joint use of AEM and RAM leads to worse post-SEO performance, and that as the legal regime is reinforced the impact of earnings management is reduced. Wu and Lo [84] indicate that institutional investors avoid management's use of REM around SEOs. We also find studies that analyse whether investing in environmentally friendly companies through capital raising operations is good for their equity. Chan and Walter [85] show that a "green" factor is important in explaining long-term stock return performance after seasoned equity offerings (SEO). They document positive and statistically significant excess returns for these firms, in contrast to the control samples used, which underperform.

But despite the extensive literature, to the best of our knowledge, there is no empirical study that has analysed whether performance management around SEO is related to the degree of the company's commitment to CSR. Thus, the purpose of this study is to find out whether CSR of companies decreases EM and whether such reduction is more evident based on around capital operations.

Therefore, our hypothesis is as follows:

H1. Higher CSR performance firms show lower earning management in issuing SEOs.

# 3. Research methodology

## 3.1. Earnings management estimation

To estimate total earnings management (TEM), we assume models used by Cohen and Zarowin [34] and by Chen and Hung [66]. Cohen and Zarowin [34] first introduced REM in the SEO environment, analysing accrual-based earnings management (AEM) and real earning management (REM) separately. Chen and Hung [66] applied their methodology to calculate TEM as the sum of the two components (AEM and REM) in the analysis of earnings manipulation in companies considered socially responsible. Thus, we define the total earnings management of firm *i* in fiscal year *t* (*TEM*<sub>*i*t</sub>) as follows:

$$TEM_{i,i} = DA_{i,i} - DCFO_{i,i} - DEXP_{i,i}$$
(1)

where  $DA_{i,t}$  represents discretionary accruals,  $DCFO_{i,t}$  shows the abnormal cash-flows from operations and  $DEXP_{i,t}$  indicates the abnormal operating expenses. We multiply both abnormal operating cash flow and abnormal operating expenses by minus one, and then add them, following the model proposed by Chen and Hung [66]. The higher DFCO and DEXP amounts, the more likely it is that the company is manipulating sales and cutting discretionary expenses to manage reported earnings upwards [34].

We estimate the discretionary accruals using the regression analysis proposed by Kothari et al. [5]:

$$\frac{TA_{i,t}}{Assets_{i,t-1}} = \alpha_0 + \alpha_1 \cdot \frac{1}{Assets_{i,t-1}} + \alpha_2 \cdot \frac{\Delta Sales_{i,t}}{Assets_{i,t-1}} + \alpha_3 \cdot \frac{TDA_{i,t}}{Assets_{i,t-1}} + \alpha_4 \cdot RoA_{i,t} + \sum_{t=1}^T \delta_t \cdot D_t + \varepsilon_{i,t}^{TA}$$
(2)

where, *TA* represents the total accruals defined as income from continuing operation minus the cash flow from operations; *Assets* is the total assets;  $\Delta$ *Sales* is the difference in sales between two consecutive fiscal years or sales growth; *TDA* shows the total depreciable assets and *RoA* denotes the total RoA (Return on Assets).  $D_t$  is a dummy for each year and shows temporary effects. Finally,  $\alpha_0$  is the constant and represents the individual effect for each firm. The residual term is  $\varepsilon_{tA}^{TA}$ .

Note that Eq. (2) is a panel data model and then, once the parameters are estimated, we calculate the nondiscretionary accruals (*NDA*) as follows:

$$NDA_{i,t} = \widehat{\alpha}_0 + \widehat{\alpha}_1 \cdot \frac{1}{Assets_{i,t-1}} + \widehat{\alpha}_2 \cdot \frac{\Delta Sales_{i,t}}{Assets_{i,t-1}} + \widehat{\alpha}_3 \cdot \frac{TDA_{i,t}}{Assets_{i,t-1}} + \widehat{\alpha}_4 \cdot RoA_{i,t} + \sum_{t=1}^{I} \widehat{\delta}_t \cdot D_t$$
(3)

And then, we estimated the discretionary accruals (*DA*) as:  $DA_{i,t} = \frac{TA_{i}}{Assets_{i,t-1}} - NDA_{i,t} = \varepsilon_{i,t}^{TA}$ . We consider that our nondiscretionary accruals are independent of temporal and individual effects since, these effects are included in Eq. (3).

Similarly, and following the proposals of [4,66], we model the operative cash-flows (CFO):

$$\frac{CFO_{i,t}}{Assets_{i,t-1}} = \beta_0 + \beta_1 \cdot \frac{1}{Assets_{i,t-1}} + \beta_2 \cdot \frac{Sales_{i,t}}{Assets_{i,t-1}} + \beta_3 \cdot \frac{\Delta Sales_{i,t}}{Assets_{i,t-1}} + \sum_{t=1}^T \delta_t \cdot D_t + \varepsilon_{i,t}^{CFO}$$

$$\tag{4}$$

where, *CFO* represents the operating cash-flows; *Sales* are those for the fiscal year and  $\varepsilon_{i,t}^{CFO}$  is the residual term. As above and from estimated parameters, we determine the normal operating cash-flows (*NCFO*) as follows:

M. González-Sánchez et al.

$$NCFO_{i,t} = \widehat{\beta}_0 + \widehat{\beta}_1 \cdot \frac{1}{Assets_{i,t-1}} + \widehat{\beta}_2 \cdot \frac{Sales_{i,t}}{Assets_{i,t-1}} + \widehat{\beta}_3 \cdot \frac{\Delta Sales_{i,t}}{Assets_{i,t-1}} + \sum_{t=1}^T \widehat{\delta}_t \cdot D_t$$
(5)

And then, discretionary operating cash-flows (*DCFO*) are defined as:  $DCFO_{i,t} = \frac{CFO_{i,t}}{Assets_{i,t-1}} - NCFO_{i,t} = \varepsilon_{i,t}^{CFO}$ . We consider that nondiscretionary operating cash-flows are independent of temporal and individual effects since these effects are included in Eq. (5).

Finally, we model operating expenses (*EXP*) as [34] to avoid the problem of modelling expenses as a function of current sales, when firms record sales upwardly to increase earnings:

$$\frac{EXP_{i,t}}{Assets_{i,t-1}} = \gamma_0 + \gamma_1 \cdot \frac{1}{Assets_{i,t-1}} + \gamma_2 \cdot \frac{Sales_{i,t-1}}{Assets_{i,t-1}} + \sum_{t=1}^T \delta_t \cdot D_t + \varepsilon_{i,t}^{EXP}$$
(6)

Where *EXP* represents the difference between operating revenues and operating income (EBIT); Sales is the revenue for fiscal year t-1 and  $\varepsilon_{i,t}^{EXP}$  is the residual. Again, we estimate normal operating expenses (*NEXP*) as:

$$NEXP_{i,t} = \hat{\gamma}_0 + \hat{\gamma}_1 \cdot \frac{1}{Assets_{i,t-1}} + \hat{\gamma}_2 \cdot \frac{Sales_{i,t-1}}{Assets_{i,t-1}} + \sum_{t=1}^T \hat{\delta}_t \cdot D_t$$

$$\tag{7}$$

And then, discretionary operating expenses (*DEXP*) are:  $DEXP_{i,t} = \frac{EXP_{i,t}}{Assets_{i,t-1}} - NEXP_{i,t} = \varepsilon_{i,t}^{EXP}$ . We consider that nondiscretionary operating expenses are independent of temporal and individual effects since these effects are included in Eq. (7).

# 3.2. Testing hypothesis

Then, since our nondiscretionary estimates are independent of temporary and individual effects, we estimate Eq. (8) to test our hypothesis:

$$TEM_{i,t} = \omega_1 \cdot D_{i,t}^{SEO} + \omega_2 \cdot D_{i,t}^{CSR} + \omega_3 \cdot D_{i,t}^{FRQ} + \delta_1 \cdot \ln(Assets_{i,t}) + \delta_2 \cdot RoA_{i,t} + \delta_3 \cdot BtM_{i,t} + \delta_4 \cdot DEBT_{i,t} + \varepsilon_{i,t}^{TEM}$$

$$\tag{8}$$

We define the dummies variables:  $D_{i,t}^{SEO}$ ,  $D_{i,t}^{CSR}$  and  $D_{i,t}^{FRQ}$ :

- $D_{i,t}^{SEO}$  is 1 when the company has increased capital in the year or in any of the 2 subsequent years (t, t+1 and t+2); otherwise, it takes the value 0.
- $D_{i,t}^{CSR}$  shows the degree of the firm's involvement with CSR. There are different ways of assigning a value to this variable in the literature, ranging from using external databases [62,86,87] to taking into consideration whether the company is included in sustainability indeces [59,88]. In any case, these ways of acting are not exempt from a certain degree of subjectivity, so some studies have constructed *ad-hoc* indicators [44] even though they do not completely eliminate the subjectivity of the measure. More recently, following the mandatory requirement in some countries for listed companies to present non-financial information within financial statements, the measurement has been based on the quality of such non-financial information [89]. In our case, we will take two measures of the CSR variable: on the one hand, the inclusion of the company in a sustainability index and, on the other, the degree of quality of the non-financial information provided by the firm according to the standards of the Global Reporting Initiative (GRI). In this way, we can contrast whether the results are sensitive to the way of measuring the degree of CRS involvement of the company. We use the Down Jones Sustainability index as other authors such as [90–95].

Thus, we consider socially responsible companies that belong to the index, so that the variable  $D_{i,t}^{CSR}$  takes value 1 if the company is part of the index in the year, and -1 if it is not included. This dummy is centered and thus, we ensure that the constant is zero, since the dependent variable, by definition, has zero mean, since the models for estimating the earnings management components already include the time and individual effects.

In addition, in order to make the analysis more robust, we complete the CSR measurement by considering whether companies prepare their sustainability reports in accordance with the criteria established by GRI, following authors such as [96–100]. GRI publishes a list of international companies that prepare reports with non-financial information according to the content standard applied, and also according to whether this information has been verified by a specialised external agent.

Thus, our variable  $D_{i,t}^{CSR}$  takes a value of 1 if the company complies with GRI standards and submits the information to external verification; we assign it a value of 0 if it complies with GRI standards but does not undergo external verification, and finally it has a value of -1 if it fails to comply with GRI standards.

- Finally, we define  $D_{i,t}^{FRQ} = D_{i,t}^{SEO} \cdot D_{i,t}^{CSR}$  as multiplicative variable. So,  $D_{i,t}^{FRQ}$  takes a value of 1 if the company has increased capital and has the highest rating of CSR; it takes a value of -1 if the company has increased capital but has the lowest rating of CSR, and otherwise it takes 0 value. Note that this is possible since the dummy variable  $D_{i,t}^{CSR}$  is centered.

Furthermore, as in the literature reviewed, we include the usual control variables of size (*ln(Assets*)), return on assets (*RoA*), growth

options measured by Book-to-Market ratio (*BtM*) and level of indebtedness measured by Debt-to-Book value ratio (*DEBT*). In short, according to estimating results of Eq. (8):

- If  $\omega_1 = 0$ , firms that increase capital do not show higher levels of EM. However, if  $\omega_1 \neq 0$ , the companies that increase capital show higher ( $\omega_1 > 0$ ) or lower levels ( $\omega_1 < 0$ ) of EM.
- If  $\omega_2 = 0$ , EM is independent of firms' CSR. But, if  $\omega_2 < 0$ , firms with higher CSR quality present lower levels of EM.
- If  $\omega_3 = 0$ , firms with higher accomplishment of CSR and that increase capital do not present different EM from the rest. Instead, if  $\omega_3 < 0$ , the firms with higher CSR quality and that increased capital show lower levels of EM around SEOs. In the latter case, we accept our hypothesis and then, when a capital increase occurs, the financial reporting quality of companies with higher CSR development is higher than that of companies with lower CSR.

## 3.3. Model estimation

For the estimation<sup>1</sup> of Eqs. (2), (4) and (6), in order to extract the residual or components of the total manipulation of the result, and given that we are dealing with a panel of data, first we determine whether the individual effect of each company is fixed (constant for the whole sample period) or random. To do so, we will use the Hausman test whose null hypothesis is that the estimation by generalised least squares (random effects) is efficient. On the contrary, if the hypothesis is rejected (p-value<0.05), then the ideal estimation procedure would be ordinary least squares with fixed effects, i.e. substituting  $\alpha_0$  by  $\alpha_{0,i}$  or including a dummy for each of the companies in the sample.

In addition, to test for the absence of endogeneity in the model, we estimate the [107] autocorrelation tests on the residuals. Thus, if the null hypothesis of absence of autocorrelation is not accepted, we should estimate the models as if they were a dynamic panel data using the generalised method of moments and using instrumental variables to correct for the endogeneity problem. We will apply this same methodology to estimate Eq. (8), since the dependent variable is the result of the previous estimations.

# 4. Data

The sample is made up of French, German, Italian and Spanish<sup>2</sup> (representing the largest EU economies) non-financial listed firms (banks, insurance, financial and real estate companies are excluded), since their financial statements are formulated in the same currency (euro) and subject to European Directives and IFRS. The data period runs from 2012 to 2020, but as some variables are estimated on the basis of differences between two consecutive fiscal years, the analysis period covers 2011–2020. To obtain all financial data, we have used Amadeus database of Bureau van Dijk (BvD). If the necessary variables are not available for one year, we exclude this company for this year. As usual in literature, we also exclude, for each firm, those years with negative book value,<sup>3</sup> as well as those extreme values that lie in the distribution tails (less than 1% or more than 99%). As a result of the search, the final sample has the following composition:

Table-1 shows a regular growth in the total number of companies in the sample, up to 2019. In 2020 there is a decrease of 21.57% in the total sample, probably as a consequence of the pandemic (COVID-19). In addition, France and Germany account for more than twice as many as Italy and Spain.

Table-2 shows the number of companies that have increased capital for each of the countries and years in the sample. Note that the country with the highest number of SEOs was France (47.7% of the total), followed by Germany with 26.8%, while the sum of the data for Italy and Spain is similar to that of Germany. On the other hand, 2014–2015 was the period with the highest percentage of SEOs per total companies in France, 2013–2014 in Germany and 2017-18 in Italy and Spain. In summary, this heterogeneity in size and periods justifies this comparative study.

<sup>&</sup>lt;sup>1</sup> To test the proposed hypothesis, we can either estimate a single model for the set of countries, or we can estimate the model for each country separately. We have opted for the latter possibility for several reasons: When the estimation is joint, we need to include a dummy variable for each country in the sample, so the number of dummy variables and their multiplicative effects grows exponentially (remember that in addition to the country dummy, we have defined those corresponding to the following effects: temporary, individual, CSR and SEO), with the consequent increase in multicollinearity problems and the added difficulty of correctly understanding multiplicative effects, for example when analyzing the multiplicative effect of year and country. On the other hand, as the literature shows [101–106], although the countries belong to a common EU area, the financial characteristics of the companies are different -and so are their capital markets-, among others: level of internationalization of companies, level of investor protection, bank-oriented versus market-oriented financial system, taxation and creditors protection. And then, including a country (or group) effect could be capturing other differentiating characteristics that could have distorting effects on the objective sought in the study.

<sup>&</sup>lt;sup>2</sup> Specifically, the initial goal was to include more EU countries (Finland and Netherlands), but the size of their capital market meant that the number of companies was low, furthermore, within that set, the number of firms included in the CSR indeces was even lower, and finally, there were almost no companies that had performed SEO in the sample period and belonged to the above subsets (for example, of the observations in Finnish companies, only 7.5% raised capital). Thus, the inclusion in our sample of a small amount of data from another country, not only seems irrelevant, but the results could be interpreted more as an idiosyncratic case than as a general behavior.

 $<sup>^{3}</sup>$  In the literature it is usual to exclude firms with negative book value for different reasons: difficult interpretation of the results or high credit risk, among others. There are usually few data, and therefore, the complexity of their treatment in isolation from the rest of the sample means that they are normally excluded. For example, in the capital market, the field in which we work in our study of SEOs, the size factor is key and these outliers are excluded for their estimation (see Refs. [108–110], among others).

Sample composition.

Year	France	Germany	Italy	Spain	Total sample
2012	497	426	197	135	1255
2013	531	439	202	147	1319
2014	552	413	219	151	1335
2015	568	405	238	180	1391
2016	589	424	249	191	1453
2017	609	427	261	213	1510
2018	619	466	286	230	1601
2019	621	489	291	222	1623
2020	557	424	214	140	1335
Sum	5143	3913	2157	1609	12822

Note: This table shows the number of companies in the sample, disaggregated by year and country.

Table 2	
Number of companies with	SEO.

Year	France	Germany	Italy	Spain	Total
2012	176	107	43	26	352
2013	177	120	34	30	361
2014	213	128	56	38	435
2015	241	124	65	53	483
2016	212	107	61	56	436
2017	232	120	82	72	506
2018	233	128	90	86	537
2019	228	122	87	57	494
2020	210	126	64	23	423
Total	1922	1082	582	441	4027

Note: This table shows the number of companies that have increased their capital for each of the countries and years in the sample.

Table-3 shows the main quartiles of the financial variables used in the empirical study after discarding outliers (below 1% or above 99%). We present quartiles instead of other usual statistics (mean and standard deviation) because the probability distribution of the variables does not have to be Gaussian, and quartiles show more accurately the dispersion of the variables throughout the entire distribution, not only around the mean (all financial variables are in thousands of euros).

Analysing the data by median, we find that BtM and size are similar for all countries. On the other hand, for the remaining variables, except total depreciable assets and debts, the lowest median value corresponds to Spanish companies, while the highest is for French (inverse of total assets), German (sales, RoA, operating cash-flows and expenses) and Italian companies (total accruals and sales growth). Finally, Spanish companies show the highest median value for debts, and German companies the lowest, while the highest and lowest values for total depreciable assets correspond to Italian and France companies, respectively.

According to the amplitude (max. minus min.), French companies show the highest dispersion in profitability (RoA) and operating cash-flow indicators; German companies show the greatest dispersion in sales and expenses indicators; Italian companies present the highest amplitude in total accruals, inverse of total assets, sales growth and size; Spanish companies show the greatest dispersion in Book-to-Market, debts, and total depreciable assets.

With regard to the CSR variable, it should be noted that our sample begins in 2012, since the Dow Jones Sustainability Index for Europe<sup>4</sup> was created in April 2010. Finally, we also consider the GRI<sup>5</sup> level of the different companies in the sample. Table-4 presents the number of companies per year and country that are included in the DJSI index (Panel A), and the number of companies whose non-financial information included in the annual financial statements (Panel B) has been externally audited (Audit) or has not been externally audited (No Audit).

In Table-4 Panel-A, we observe that around 3% of the total number of companies in France, Germany and Italy are included in the DJSI index, while in Spain this percentage rises to 5.3%. Table-4 Panel-B shows that the percentage of companies whose non-financial CSR information has been externally verified is higher than those without external verification. By country and overall, Spanish companies show the highest percentage of externally verified information, while French companies show the lowest Finally, note that the percentage of companies with and without external verification is 6.35% for the total sample (12822 observations), while the percentage of companies included in the DJSI index is 3.31%, so that the GRI seems to be less restrictive than the DSJI index and, as a consequence, the differentiating analysis of both indeces is justified for our aim.

<sup>&</sup>lt;sup>4</sup> The inclusion of the companies in this index can be consulted at: https://www.spglobal.com.

<sup>&</sup>lt;sup>5</sup> GRI database has been consulted at https://database.globalreporting.org/(available until September 2021) and https://www.globalreporting.org/reportregistration/verifiedreports (available from October 2021).

Table 3		
Quartile an	alysis of financial variabl	es.

m 11 o

Variable	Quartile	France	Germany	Italy	Spain	Variable	Quartile	France	Germany	Italy	Spain
$TA_{i,t}$	Max	-0.0001	0.0000	-0.0007	0.0000	CFO <sub>i.t</sub>	Max	0.4022	0.5507	0.4265	0.5080
$Assets_{i,t-1}$	Upper quartile	-0.0166	-0.0201	-0.0278	-0.0123	Assets <sub>i.t-1</sub>	Upper quartile	0.0949	0.1167	0.1175	0.0984
	Median	-0.0333	-0.0375	-0.0449	-0.0271		Median	0.0541	0.0734	0.0695	0.0500
	Lower quartile	-0.0563	-0.0593	-0.0702	-0.0452		Lower quartile	0.0059	0.0281	0.0311	0.0056
	Min	-0.2526	-0.2406	-0.2628	-0.2176		Min	-1.0969	-0.8300	-0.4135	-0.3402
$EXP_{i.t}$	Max	3.6485	5.5128	5.2131	3.0135	100	Max	0.1754	0.2482	0.4487	0.1803
$\overline{Assets_{i.t-1}}$	Upper quartile	1.1827	1.3164	1.0641	0.7839	$Assets_{i,t-1}$	Upper quartile	0.0061	0.0035	0.0039	0.0039
	Median	0.8043	0.8760	0.7373	0.4149		Median	0.0011	0.0007	0.0007	0.0006
	Lower quartile	0.4668	0.4467	0.4720	0.0559		Lower quartile	0.0001	0.0001	0.0002	0.0001
	Min	0.0073	0.0075	0.0096	0.0038		Min	0.0000	0.0000	0.0000	0.0000
$TDA_{i,t}$	Max	1.1755	1.3280	1.5638	1.5738	$\Delta Sales_{i,t}$	Max	1.1704	1.4803	2.4265	0.9237
$Assets_{i,t-1}$	Upper quartile	0.5165	0.5492	0.5849	0.5920	Assets <sub>i,t-1</sub>	Upper quartile	0.0883	0.1039	0.1265	0.0507
	Median	0.3428	0.3515	0.4162	0.3755		Median	0.0156	0.0244	0.0286	0.0053
	Lower quartile	0.1553	0.1283	0.2383	0.0818		Lower quartile	-0.0358	-0.0305	-0.0291	-0.0142
	Min	0.0000	0.0000	0.0019	0.0000		Min	-0.8932	-1.0256	-0.6419	-0.5317
$Sales_{i,t}$	Max	3.7180	5.3600	5.0486	3.0616	$Sales_{i:t-1}$	Max	4.1466	4.5860	2.9356	2.7092
$Assets_{i,t-1}$	Upper quartile	1.1990	1.3846	1.1542	0.8150	Assets <sub>i.t-1</sub>	Upper quartile	1.1602	1.3137	1.0391	0.8057
	Median	0.7861	0.9179	0.7783	0.4194		Median	0.7777	0.8996	0.7401	0.4338
	Lower quartile	0.4055	0.4302	0.4835	0.0730		Lower quartile	0.4030	0.4294	0.4795	0.0652
	Min	0.0001	0.0019	0.0021	0.0001		Min	0.0000	0.0018	0.0000	0.0001
$RoA_{i,t}$	Max	0.5282	0.4824	0.6568	0.3298	$DEBT_{i.t}$	Max	8.8388	6.7813	12.3630	25.8177
	Upper quartile	0.0801	0.0952	0.0832	0.0679		Upper quartile	0.9637	1.0043	1.4760	1.6707
	Median	0.0368	0.0487	0.0425	0.0314		Median	0.4713	0.4603	0.7582	0.8131
	Lower quartile	-0.0235	-0.0043	-0.0014	-0.0065		Lower quartile	0.1710	0.1405	0.3506	0.3583
	Min	-1.1805	-1.0368	-0.4746	-0.3240		Min	0.0000	0.0000	0.0000	0.0000
$BtM_{i.t}$	Max	5.9057	4.4384	5.4987	36.3510	$ln(Assets_{i.t})$	Max	18.3281	18.5000	18.5128	18.3418
	Upper quartile	1.1400	0.9945	1.1778	1.1670		Upper quartile	13.4782	13.5465	13.3459	14.1391
	Median	0.6728	0.6039	0.6813	0.6643		Median	11.4688	11.8897	11.9211	11.9847
	Lower quartile	0.3742	0.3371	0.3798	0.3530		Lower quartile	9.7676	10.3309	10.2341	10.2673
	Min	0.0315	0.0477	0.0554	0.0434		Min	6.3572	6.1321	5.9873	6.9430

Note: This table shows the main quartiles of the financial variables used disaggregated by country. Values are shown for the variables in Eqs (2), (4) and (6), i.e.: *TA*: Total accruals defined as income from continuing operation minus the cash flow from operations; *CFO*: Operating cash-flows; *Sales*: Revenue; *EXP*: Difference between operating revenues and operating income (EBIT); *Assets*: Total assets; *ASales*: Difference in revenue between two consecutive fiscal years or sales growth; *TDA*: Total depreciable assets; *RoA*: Return on Assets; *Ln(Assets)*: Log-asset; *BtM*: Book-to-Market ratio; *DEBT*: Level of indebtedness measured by Debt-to-Book value ratio.

Analysis of dummies for CSR.

Panel A. DJSI	Index companies							
Year	France		Germany		Italy		Spain	
2012	12		13		8		13	
2013	15		19		8		11	
2014	14		16		8		8	
2015	19		11		7		9	
2016	22		9		6		10	
2017	24		9		6		9	
2018	24		9		4		9	
2019	22		10		5		9	
2020	21		12		6		8	
Sum	173		108		58		86	
Panel B. Com	panies by GRI rat	ing						
Year	Audit	No Audit	Audit	No Audit	Audit	No Audit	Audit	No Audit
2012	9	3	20	12	11	3	22	1
2013	8	1	20	15	13	2	22	1
2014	10	3	18	13	13	1	23	4
2015	9	0	20	11	12	0	23	5
2016	7	0	22	8	13	0	24	3
2017	9	0	35	8	19	0	25	4
2018	8	0	40	6	20	0	28	1
2019	9	0	42	6	20	0	27	2
2020	8	0	39	6	20	0	27	0

Note: This table shows number of companies included in the DJSI Index (Panel A) and in the GRI rating (Panel B) disaggregated by year. In Panel B we distinguish between the number of companies that have been externally verified and those that have not been externally verified.

### 5. Results and discussion

#### 5.1. Earnings management results

First, we estimate the explanatory models of total accruals, operating cash-flows and operating expenses according to Eqs. (2), (4) and (6), in order to determine, from the values obtained, the discretionary accruals, operating cash-flows and expenses. Table-5 shows the results.

From the results in Table-5 we see that the Hausman test rejects the hypothesis on the consistency of the GLS estimation in all cases, so that the individual effects included in the model are fixed and the estimation method LSDV, while the standard errors are robustly estimated in the face of heteroscedasticity and autocorrelation. We observe that the explanatory power is very similar across countries. However, there are slight differences for each dependent variable. Finally, it should be noted that the [107] autocorrelation tests on the residuals (AR (1) and AR (2)) show the absence of time dependence of the residuals, so that from these results we consider that the models do not show endogeneity.

From the results analysed in Table-5, we can see that although the models used are statistically significant, the values obtained are different by country, which justifies our analysis of the results management differentiated by the origin of the company. So, for total accruals (Eq. (2)), the inverse of total assets is only significant for Spanish companies (with negative impact). Sales growth in Spanish companies is not significant; for the rest of the countries the effect is negative, with Italian companies showing the highest weight. The effect of depreciable assets is negative and significant in all countries, although the weight is slightly lower for French companies. Finally, RoA shows a significant positive effect for all countries, although Italian companies show a weight of more than double that of companies in the other countries, consistently with [45,111]. For operating cash flows (Eq. (4)), we find that the inverse of total assets has only a significant negative effect for German companies. With respect to sales growth, the effect is positive and Spanish companies have the highest weight for this variable. Lastly, and according to [4,112], fiscal year sales also show a positive and significant effect for all countries, but France and Spanish companies present the highest weight for this variable. Finally, in the operating expenses model (Eq. (6)), in line with [4,49,112], we find that the two regressors show positive and statistically significant parameters for all countries and, while sales in the previous fiscal year have a similar weight on the operating expenses for all companies, the inverse of total assets shows a much higher effect in the case of French companies.

Table-6 shows an analysis by quartiles of discretionary accruals, operating cash-flows and expenses for each country.

Discretionary accruals and operating cash-flows show a clear positive asymmetry as the lowest quartile show higher values in absolute value than the highest quartile. The opposite is true for discretionary expenses in all countries in the sample. Moreover, discretionary accruals show lower values than discretionary cash-flows for the respective percentiles, and the latter show lower values than discretionary expenses. This is relevant and justifies the analysis of each component of TEM because, although the estimation model for discretionary expenses showed the highest explanatory power according to the adjusted R2 (see Table-5), we now see in Table-6 that these are the ones that show the highest values and therefore the widest range of values.

Table 5
Results from estimates of discretionary accruals, operating cash-flows, and expenses.

Regressors	$\frac{TA_{i,t}}{Assets_{i,t-1}} (2)$		$\frac{CFO_{i,t}}{Assets_{i,t-1}} $ (4)		$\frac{EXP_{i,t}}{Assets_{i,t-1}}$ (6)		$\frac{TA_{i,t}}{Assets_{i,t-1}} (2)$		$\frac{CFO_{i,t}}{Assets_{i,t-1}} $ (4)		$\frac{EXP_{i,t}}{Assets_{i,t-1}}$ (6)	
	parameter	t-value	parameter	t-value	parameter	t-value	parameter	t-value	parameter	t-value	parameter	t-value
Panel A. Results from	French compan	ies					Panel B. Res	ults from German	companies			
1	8.4106	0.683	-32.8478	-0.421	766.44	5.36 (***)	-10.4328	-1.34	-115.442	-2.3 (**)	483.711	2.95 (***)
$Assets_{i,t-1}$												
$\Delta Sales_{i,t}$	-0.0091	-3.59 (***)	0.0566	3.43 (***)			-0.0072	-2.94 (***)	0.0830	4.88 (***)		
Assets <sub>i,t-1</sub>	0.0177	0 50 (***)					0.0410	7 00 (***)				
Assets	-0.0177	-3.50 (***)					-0.0419	-7.83 (^^^)				
RoAit	0.0558	8 42 (***)					0.0369	4 59 (***)				
Sales; t	0.0000	0.12()	0.0930	7.01 (***)			0.0009	1.05()	0.0622	3.35 (***)		
$\overline{Assets_{i,t-1}}$												
$Sales_{i,t-1}$					0.5591	14.30 (***)					0.6545	10.9 (***)
$\overline{Assets_{i,t-1}}$												
Individual effects	Yes		Yes		Yes		Yes		Yes		Yes	
Temporary effects	Yes		Yes		Yes		Yes		Yes		Yes	
Hausman	87.2512	(0.000)	20.1548	(0.000)	194.8127	(0.000)	11.4852	(0.021)	35.4571	(0.000)	244.0742	(0.000)
AR (1) residuals	0.704	(0.459)	1.166	(0.243)	1.935	(0.053)	0.910	(0.363)	1.183	(0.237)	1.306	(0.192)
AR (2) residuals	0.592	(0.554)	1.082	(0.279)	0.399	(0.690)	0.894	(0.371)	1.053	(0.292)	0.686	(0.493)
adjusted R <sup>2</sup>	75.02%		77,30%		87,19%		75.78%		69.53%		91.07%	
Panel C. Results from	Italian compan	ies					Panel D. Results from Spanish companies					
1	7.0441	1.32	-44.1474	-1.64	232.073	2.39 (**)	-16.9974	-2.26 (**)	-56.5195	-0.912	380.061	1.96 (*)
$\overline{Assets_{i,t-1}}$												
$\Delta Sales_{i,t}$	-0.0171	-4.16 (***)	0.0391	2.26 (**)			0.0011	0.231	0.1218	3.74 (***)		
Assets <sub>i,t-1</sub>												
$\frac{TDA_{i,t}}{A_{i,t}}$	-0.0432	-4.95 (***)					-0.0357	-4.15 (***)				
Assets <sub>i,t-1</sub> PoA	0 1205	7 64 (***)					0.0220	1 09 (**)				
Sales.	0.1205	7.04()	0.0765	5 65 (***)			0.0229	1.98()	0.0917	4 22 (***)		
Assets			0.0703	5.05 ( )					0.0517	4.22()		
Salesi 1					0.6241	9.22 (***)					0.5725	7.14 (***)
$\overline{Assets_{i,t-1}}$					010211	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,					010/20	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,
Individual effects	Yes		Yes		Yes		Yes		Yes		Yes	
Temporary effects	Yes		Yes		Yes		Yes		Yes		Yes	
Hausman	22.1348	(0.000)	21.2591	(0.000)	228.132	(0.000)	28.8745	(0.000)	29.1146	(0.000)	188.2586	(0.000)
AR (1) residuals	1.716	(0.086)	1.098	(0.272)	1.112	(0.266)	0.995	(0.340)	1.186	(0.236)	1.123	(0.262)
AR (2) residuals	0.517	(0.605)	0.990	(0.322)	0.453	(0.650)	0.201	(0.842)	1.006	(0.315)	0.991	(0.322)
adjusted R <sup>2</sup>	76.34%		69.66%		88.84%		87.63%		63.90%		92.95%	

Note: This table shows the results of estimating discretionary accruals (according to equation (2)), operating cash flows (according to equation (4)) and operating expenses (according to equation (6)) for the French (Panel A), German (Panel B), Italian (Panel C) and Spanish (Panel D) companies. *TA*: Total accruals defined as income from continuing operation minus the cash flow from operations; *CFO*: Operating cash-flows; *Sales*: Revenue; *EXP*: Difference between operating revenues and operating income (EBIT); *Assets*: Total assets; *ASales*: Difference in revenue between two consecutive fiscal years or sales growth; *TDA*: Total depreciable assets; *RoA*: Return on Assets. Hausman test rejects the hypothesis on the consistency of the GLS estimation in all cases, the autocorrelation tests (AR(1) and AR(2)) show the absence of time dependence of the residuals and adjusted R2 indicates the explanatory power. (\*), (\*\*) and (\*\*\*) indicate that the parameter is statistically significant at a confidence level of 10%, 5% and 1%, respectively.

Quartiles of discretionary accruals abnormal operating cash-flows and abnormal expenses.

Discretionary	Quartile	German	France	Italy	Spain
Accruals	Max.	0.1219	0.1179	0.1107	0.0443
	Upper quartile	0.0067	0.0074	0.0081	0.0039
	Median	0.0006	0.0007	0.0003	0.0002
	Lower quartile	-0.0043	-0.0048	-0.0061	-0.0031
	Min.	-0.1575	-0.1750	-0.1422	-0.1128
Operating Cash-Flows	Max.	0.6003	0.5577	0.3287	0.4917
	Upper quartile	0.0220	0.0235	0.0191	0.0208
	Median	0.0000	0.0011	0.0000	0.0000
	Lower quartile	-0.0217	-0.0201	-0.0204	-0.0182
	Min.	-0.7400	-0.8133	-0.3587	-0.2803
Operating Expenses	Max.	2.8504	2.4044	2.2378	1.5964
	Upper quartile	0.0584	0.0561	0.0567	0.0363
	Median	-0.0019	-0.0057	-0.0053	0.0000
	Lower quartile	-0.0651	-0.0628	-0.0629	-0.0351
	Min.	-1.8989	-1.5073	-1.1166	-1.0016

Note: This table shows an analysis by quartiles of discretionary accruals, operating cash-flows and expenses for each country.

# 5.2. Hypothesis testing and discussion

To test our research hypothesis, we estimate Eq. (8) by applying the OLS and GLS models, and present the results in Table-7 which shows the results of the estimation using DJSI as the explanatory variable for CSR, and in Table-8 which presents the results using GRI. Furthermore, in order to draw more robust conclusions, we have estimated Eq. (8) using not only TEM as the dependent variable, but also its decomposition according to Eq. (1) into discretionary accruals (DA), discretionary operating cash-flows (DCFO) and discretionary operating expenses (DEXP).

From Tables 7 and 8, we observe that our results are robust since both the value of the parameters and their statistical significance do not show relevant variations.<sup>6</sup> Tables 7 and 8 show that when using the GRI dummy as a measure of CSR quality, the goodness-of-fit of the model is slightly higher than when using the DJSI dummy. We also observe the highest explanatory power for the discretionary operating cash-flows, while discretionary operating expenses shows the lowest level of goodness-of-fit. By country, the explanatory power from lowest to highest value and for any of the dependent variables is: Spain, Italy, Germany and France. Therefore, there is an inverse relationship between financial market size and the explanatory power of the model. With respect to the control variables, we observe that they do not show the same explanatory capacity either across countries and for each dependent variable.

The components of TEM show a different sensitivity to each control variable, and furthermore, this sensitivity is different for each country in the sample; the results of estimating Eq. (8) for the TEMs show this variety. Focusing on the control variables, in the DA model, French and German companies show a negative relationship with the level of indebtedness. In DCFO model, we observe similar behaviour in all the countries analysed when we take profitability (with a positive relationship) and firm size (with a negative relationship) as control variables. French and German companies also exhibit a similar behaviour in the DEXP model where there is a negative relationship with BtM.

Although the countries analysed use the same currency and are subject to a common basic regulatory framework, our studies reveal differences in their behaviour. Despite this common framework, each country shows different factors in relation to its institutional environment. In this context, Halabi et al. [113] identify three factors: investor protection, enforcement quality and equity market development. According to their study, French and German companies showed similar results to one another and differ from those obtained by Italian and Spanish companies.

In the model with the highest explanatory power (DCFO), for French, German and Italian companies we find a positive relationship between capital operations in subsequent years and discretionary operating cash-flows, i.e., companies in these countries increase their discretionary operating cash-flows during the two years prior to the equity issues (SEO). Given that discretionary operating cash-flows is part of the REM measurement, these results are in line with previous literature that shows real activity manipulation in the SEO environment [28,34]. Only French companies show a decrease in DCFO the better the quality and development of their CSR, proxied by both the firm's DJSI membership and its GRI level. This result is in line with [61] who observe in the French context that EM has a negative impact on financial performance and that CSR moderates this impact. Finally, the hypothesis that firms with better CSR quality and development (DJSI or GRI level) show lower operating cash-flows management in the years preceding equity issues is rejected.

We observe that for Spanish companies there is no relationship between TEM or any of its components (DA, DCFO and DEXP) with subsequent years' capital operations (SEO) or with the quality and development of CSR as measured by companies'inclusion in the DJSI. However, we found a negative and significant relationship between DA and Spanish socially responsible companies as measured by the GRI level of their non-financial information. This result is in line with that obtained for Spanish companies by Gras-Gil et al. [45]

<sup>&</sup>lt;sup>6</sup> We appreciate the comments made in the review of the paper that recommended showing the estimation results with both the OLS and GLS methodologies.

Table 7	
Results using Dow Jones Sustainability Index.	

	OLS ESTIMATION									GLS ESTIMATION						
Panel A. Fra	ıce															
Regressors	DA parameter	t-value	DCFO parameter	t-value	DEXP parameter	t-value	TEM parameter	t-value	DA parameter	t-value	DCFO parameter	t-value	DEXP parameter	t-value	TEM parameter	t-value
ROA	-0.0010	-0.5400	0.1484	7.92	0.0548	1.74 (*)	-0.1850	-6.33 (***)	-0.0001	-0.42	0.1308	6.45	0.0592	1.71	-0.1633	-6.01 (***)
BtM	-0.0003	-0.89	8.9062	0.48	-0.0089	-2.28 (**)	0.0084	2.13 (**)	-0.0004	-0.71	6.4513	0.39	-0.0182	-2.15 (**)	0.0082	2.08 (**)
DEBT	-0.0013	-3.32 (***)	-0.0033	-1.83 (*)	0.0051	1.3600	-0.0026	-0.72	-0.0011	-3.08 (***)	-0.0015	-1.79 (*)	0.0055	1.29	-0.0018	-0.64
LN	0.0001	1.97 (**)	-0.0010	-2.88 (***)	-0.0002	-0.35	0.0011	1.71 (*)	0.0001	1.95	-0.0010	-2.66 (***)	-0.0002	-0.25	0.0011	1.69 (*)
SEO	-0.0011	-0.78	0.0161	3.01 (***)	0.0070	0.81	-0.0212	-2.12 (**)	-0.0026	-0.59	0.0142	2.96	0.0094	0.78	-0.0207	-2.09 (**)
DJSI	-8.7287	-0.12	-0.0087	-2.08 (**)	0.0003	0.48	0.0062	0.82	-7.4194	-0.09	-0.0228	-2.01 (**)	0.0006	0.45	0.0048	0.73
FRQ-DJSI adjusted R <sup>2</sup>	-0.0005 0.40%	-0.37	0.0032 9.28%	0.64	0.0002 0.42%	0.29	-0.0020 2.81%	-0.21	-0.0009 0.38%	-0.33	0.0014 9.16%	0.55	0.0002 0.39%	0.24	-0.0047 2.75%	-0.17
Panel B. Gerr	nany															
Regressors	DA parameter	t-value	DCFO parameter	t-value	DEXP parameter	t-value	TEM parameter	t-value	DA parameter	t-value	DCFO parameter	t-value	DEXP parameter	t-value	TEM parameter	t-value
ROA	0.0039	1.1200	0.2131	8.03	0.0348	0.5970	-0.2279	-3.30 (***)	0.0006	1.05	0.1862	7.55	0.0808	0.55	-0.2113	-3.210 (***)
BtM	-2.6958	-0.53	0.0033	1.4100	-0.0122	-1.72 (*)	0.0089	1.2400	-2.4898	-0.47	0.0097	1.36	-0.0110	-1.68 (*)	0.0018	1.19
DEBT	-0.0008	-1.68 (*)	-0.0018	-0.94	0.0012	0.27	-2.4377	-0.57	0.0009	-1.65 (*)	-0.0045	-0.89	-0.0008	0.23	-1.8283	-0.51
LN	4.3290	0.6500	-0.0013	-4.71 (***)	0.0002	0.35	0.0009	1.3800	5.5844	0.60	-0.0015	-4.36 (***)	0.0001	0.31	0.0002	1.33
SEO	-0.0018	-1.0200	0.0081	1.76 (*)	0.0010	0.11	-0.0099	-1.0300	-0.0047	-0.97	0.0071	1.71 (*)	0.0002	0.08	-0.0052	-0.96
DJSI	0.0003	0.34	-0.0026	-0.98	0.0012	0.21	0.0012	0.19	0.0002	0.31	-0.0059	-0.91	0.0015	0.16	0.0011	0.14
adjusted R <sup>2</sup>	-0.0017 0.30%	-0.98	0.0019 14.78%	0.43	-0.0022 0.17%	-0.32	-0.0008 1.73%	-0.74	-0.0049 0.29%	-0.86	0.0011 14.52%	0.38	-0.0063 0.15%	-0.28	-0.0006 1.67%	-0.66
Damal C. Ital																
Regressors	DA parameter	t-value	DCFO	t-value	DEXP parameter	t-value	TEM parameter	t-value	DA parameter	t-value	DCFO	t-value	DEXP parameter	t-value	TEM parameter	t-value
ROA	0.0117	1.4200	0.1672	8.61	0.0971	1.66 (*)	-0.2348	-3.57	0.0138	1.35	0.1445	8.11	0.1039	1.66	-0.2664	-3.49
BtM	0.0009	2.07 (**)	0.0035	(***) 2.77	0.0069	1.50	-0.0089	(***) -2.13	0.0010	2.01	0.0032	(***) 2.52	0.0084	(*) 1.40	-0.0096	(***) -2.06
DEBT	-0.0003	-1.1600	0.0020	(***) 2.66 (**)	-0.0014	-0.58	-0.0006	(**) -0.2700	-0.0003	(**) -1.02	0.0025	(***) 2.44 (**)	-0.0014	-0.51	-0.0007	(**) -0.21
LN	0.0000	-0.21	-0.0010	-4.68 (***)	-0.0006	-1.01	0.0014	2.28 (**)	0.0000	-0.15	-0.0010	(**) -4.35 (***)	-0.0008	-0.96	0.0017	2.23 (**)
															(continued or	n next page)

Table 7 (continued)

	OLS ESTIM	ATION							GLS ESTIMATION							
SEO	-0.0023	-0.63	0.0051	1.97 (**)	-0.0024	-0.31	-0.0046	-0.51	-0.0020	-0.55	0.0060	1.95 (**)	-0.0022	-0.26	-0.0052	-0.44
DJSI	-0.0001	-0.78	0.0008	0.39	0.0062	1.0400	-0.0071	-1.21	-0.0001	-0.71	0.0008	0.31	0.0077	1.02	-0.0079	-1.16
FRQ-DJSI	-0.0009	-0.24	-0.0007	-0.24	-0.0118	-1.4700	0.0117	1.2600	-0.0008	-0.14	-0.0008	-0.18	-0.0097	-1.41	0.0152	1.21
adjusted R <sup>2</sup>	0.84%		10.79%		0.48%		1.98%		0.77%		10.49%		0.44%		1.89	
Panel D. Spa	in															
Regressors	DA		DCFO		DEXP		TEM		DA		DCFO		DEXP		TEM	
	parameter	t-value	parameter	t-value	parameter	t-value	parameter	t-value	parameter	t-value	parameter	t-value	parameter	t-value	parameter	t-value
ROA	0.0008	0.17	0.1727	4.49 (***)	0.0085	0.12	-0.1740	-2.27 (**)	0.0008	0.12	0.1788	4.21 (***)	0.0103	0.09	-0.2227	-2.11Ç (**)
BtM	0.0000	0.14	0.0008	0.6920	0.0000	0.21	-0.0008	-0.81	0.0000	0.10	0.0008	0.61	0.0000	0.18	-0.0010	-0.76
DEBT	0.0000	-0.56	-0.0007	-0.7830	-0.0009	-0.61	0.0015	0.97	0.0000	-0.51	-0.0006	-0.70	-0.0008	-0.56	0.0016	0.92
LN	0.0000	-0.5100	-0.0004	-2.03	0.0000	0.38	0.0004	0.82	0.0000	-0.44	-0.0005	-1.98	0.0000	0.32	0.0005	0.76
				(**)								(**)				
SEO	0.0008	0.67	-0.0046	-1.0200	0.0049	0.44	0.0007	0.64	0.0010	0.58	-0.0044	-0.96	0.0056	0.41	0.0006	0.59
DJSI	-0.0007	-1.41	0.0014	0.6010	0.0010	0.29	-0.0029	-0.79	-0.0007	-1.36	0.0014	0.55	0.0012	0.22	-0.0025	-0.73
FRQ-DJSI	0.0010	0.79	-0.0069	-1.5700	0.0026	0.22	0.0051	0.46	0.0011	0.72	-0.0063	-1.52	0.0026	0.16	0.0063	0.38
adjusted R <sup>2</sup>	0.14%		7.91%		0.10%		1.91%		0.13%		7.74%		0.09%		1.87%	

**Note:** This table shows the results of the estimation of Eq (8) using the DJSI as explanatory variable disaggregated by country: France (Panel A), Germany (Panel B), Italy (Panel C) and Spain (Panel D). Dependent variable: TEM (total earnings management). Disaggregation of TEM: DA (discretionary accruals), DCFO (discretionary cash flow from operations), DEXP (discretionary operating expenses). Control variables: ROA: Return on Assets; BtM: Book-to-Market ratio; DEBT: Level of indebtedness measured by Debt-to-Book value ratio; LN: Log-asset; Dummy variables: SEO: 1 when the company has increased capital in t, t+1 and t+2; otherwise, 0; DJSI: 1 if the company is part of the index in the year, and -1 if it is not included; FRQ\_DJSI: Multiplicative variable that takes the value of 1 if the company has increased the capital and has the highest rating of DJSI; takes a value of -1 if the company has increased the capital but has the lowest rating of DJSI, and otherwise takes 0 value. (\*), (\*\*) and (\*\*\*) indicate that the parameter is statistically significant at a confidence level of 10%, 5% and 1%, respectively. Adjusted R2 indicates that the explanatory power of the model. GLS and OLS estimation results are provided.

# Results using GRI.

	OLS ESTIMATION									GLS ESTIMATION								
Panel A. Fran Regressors	nce DA parameter	t-value	DCFO parameter	t-value	DEXP parameter	t-value	TEM parameter	t-value	DA parameter	t-value	DCFO parameter	t-value	DEXP parameter	t-value	TEM parameter	t-value		
ROA	-0.00128	-0.66	0.15338	8.09	0,05400	1,71 (*)	-0,18848	-6.44 (***)	-0.00108	-0,53	0,14986	7,88	0,05221	1,68 (*)	-0,20412	-6,37 (***)		
BtM	-0.00028	-0.74	-0.00080	-0.43	-0,00870	-2,21 (**)	0,00907	2.26 (**)	-0.00035	-0,71	-0,00072	-0,41	-0,00723	-2,17 (**)	0,00889	2,19 (**)		
DEBTS	-0.00126	-3.35 (***)	-0.00313	-1.74 (*)	0,00505	1,35	-0,00274	-0.75	-0.00133	-3,12 (***)	-0,00311	-1,71 (*)	0,00600	1,25	-0,00301	-0,73		
LN	0.00017	2.86 (***)	-0.00212	-5.55 (***)	0,72225	0,86	0,00190	2.32 (**)	0.00022	2,53 (***)	-0,00259	-5,42 (***)	0,66374	0,82	0,00209	2,28 (**)		
SEO	-0.00176	-1.97 (**)	0.02935	5.1 (**)	0,00220	0,17	-0,02840	-2.24 (**)	-0.00169	-1,95 (**)	0,03768	4,93 (**)	0,00237	0,15	-0,03359	-2,21 (**)		
GRI	0.00053	0.72	-0.02343	-4.88 (***)	0,00320	0,31	0,01704	1.72 (*)	0.00065	0,70	-0,02807	-4,79 (***)	0,00263	0,26	0,01576	1,70 (*)		
FRQ-GRI	-0.00118	-1.33	0.01698	3.44 (***)	-0,00482	-0,39	-0,00949	-0.79	-0.00111	-1,31	0,02025	3,36 (***)	-0,00626	-0,34	-0,00829	-0,73		
adjusted R <sup>2</sup>	0.43%		9.64%		0.43%		2.84%		0,40%		9,35%		0,39%		2.72%			
Panel B. Gerr	nany																	
Regressors	DA parameter	t-value	DCFO parameter	t-value	DEXP parameter	t-value	TEM parameter	t-value	DA parameter	t-value	DCFO parameter	t-value	DEXP parameter	t-value	TEM parameter	t-value		
ROA	0.00395	1.14	0.21303	8.03	0,03534	0,61	-0,22832	-3.31	0.00504	1,09	0,27588	7,88	0,04199	0,58	-0,27810	-3,28		
BtM	0.00010	0.19	0.00317	1.33	-0,01161	-1,66 (*)	0,00857	1.21	0.00009	0,12	0,00298	1,29	-0,01132	-1,65 (*)	0,00957	1,18		
DEBTS	-0.00074	-1.69 (*)	-0.00189	-0.97	0,00125	0,28	-0,00002	-0.53	-0.00093	-1,67 (*)	-0,00226	-0,92	0,00135	0,26	-0,00002	-0,51		
LN	0.00006	1.21	-0.00121	-5.22 (***)	0,00039	0,61	0,00077	1.14	0.00007	1,18	-0,00114	-5,17 (***)	0,00043	0,57	0,00072	1,13		
SEO	-0.00124	-1.76 (*)	0.00746	2.96 (***)	-0,00271	-0,44	-0,00529	-0.85	-0.00103	-1,72 (*)	0,00908	2,91 (***)	-0,00254	-0,42	-0,00685	-0,81		
GRI FRQ-GRI	0.00072 -0.00134	1.75 (*) -1.77	-0.00255 0.00139	-1.49 0.56	0,00460 -0,00740	$1,22 \\ -1,32$	-0,00164 0,00516	-0.41 0.89	$0.00083 \\ -0.00142$	1,71 (*) -1,73	-0,00263 0,00163	-1,43 0,52	0,00509 -0,00766	$1,21 \\ -1,29$	-0,00195 0,00472	-0,36 0,85		
adjusted R <sup>2</sup>	0.31%	(^)	14.81%		0.19%		1.75%		0,28%	(^)	14,42%		0,18%		1.63%			
Panel C. Italy	/																	
Regressors	DA		DCFO		DEXP		TEM		DA		DCFO		DEXP		TEM			
	parameter	t-value	parameter	t-value	parameter	t-value	parameter	t-value	parameter	t-value	parameter	t-value	parameter	t-value	parameter	t-value		
ROA	0.01176	1.41	0.16704	8.59 (***)	0,09789	1,67 (*)	-0,23532	-3.57 (***)	0.01126	1,39	0,17172	8,51 (***)	0,08085	1,64 (*)	-0,20285	-3,49 (***)		
BtM	0.00088	1.99 (**)	0.00344	2.71 (***)	0,00693	1,51	-0,00892	-2.13 (**)	0.00072	1,96 (*)	0,00321	2,68 (***)	0,00620	1,48	-0,00845	-2,09 (**)		
DEBTS	-0.00031	-1.17	0.00199	2.64 (***)	-0,00132	-0,53	-0,00072	-0.31	-0.00025	-1,09	0,00200	2,57 (***)	-0,00114	-0,49	-0,00063	-0,28		

(continued on next page)

Table 8 (continued)

	OLS ESTIMATION									GLS ESTIMATION								
LN	-0.00005	-0.75	-0.00099	-5.77 (***)	-0,00079	-1,31	0,00161	2.64 (**)	-0.00006	-0,72	-0,00098	-5,69 (***)	-0,00101	-1,28	0,00198	2,55 (**)		
SEO	-0.00055	-0.38	0.00377	1.96 (**)	0,00043	0,73	-0,00447	-0.47	-0.00065	-0,35	0,00332	1,91 (*)	0,00036	0,67	-0,00387	-0,42		
GRI	-0.00061	-1.67 (*)	0.00006	0.54	0,00485	1,22	-0,00541	-1.42	-0.00054	-1,65 (*)	0,00006	0,51	0,00523	1,19	-0,00556	-1,38		
FRQ-GRI adjusted R <sup>2</sup>	0.00113 0.86%	0.84	-0.00205 10.81%	-0.81	-0,00973 0.51%	-1,28	0,01289 2.01%	1.38	0.00104 0,84%	0,83	-0,00235 10,73%	-0,77	-0,01036 0,48%	-1,24	0,01610 1.84%	1,32		
Panel D. Spain																		
Regressors	DA		DCFO		DEXP		TEM		DA		DCFO		DEXP		TEM			
	parameter	t-value	parameter	t-value	parameter	t-value	parameter	t-value	parameter	t-value	parameter	t-value	parameter	t-value	parameter	t-value		
ROA	0.00048	0.09	0.17249	4.48 (***)	0,00744	0,11	-0,17328	-2.26 (**)	0.00044	0,03	0,16248	4,42 (***)	0,00738	0,10	-0,18038	-2,21 (**)		
BtM	0.00003	0.11	0.00083	0.69	0,00004	0,24	-0,00087	-0.82	0.00004	0,08	0,00094	0,62	0,00003	0,22	-0,00101	-0,78		
DEBTS	-0.00005	-0.19	-0.00060	-0.64	-0,00109	-0,72	0,00159	0.98	-0.00005	-0,14	-0,00071	-0,60	-0,00096	-0,69	0,00205	0,93		
LN	-0.00001	-0.17	-0.00050	-2.62 (***)	-0,00004	-0,94	0,00051	1.16	-0.00001	-0,12	-0,00054	-2,57 (***)	-0,00005	-0,91	0,00058	1,12		
SEO	0.00094	1.11	-0.00237	-0.71	0,00902	1,12	-0,00540	-0.68	0.00092	1,06	-0,00301	-0,66	0,00791	1,08	-0,00524	-0,64		
GRI	-0.00079	-1.67 (*)	0.00016	0.87	0,00037	0,10	-0,00111	-0.32	-0.00096	-1,65 (*)	0,00016	0,85	0,00042	0,09	-0,00095	-0,28		
FRQ-GRI	0.00172	1.88 (*)	-0.00585	-1.6	0,01010	1,21	-0,00270	-0.33	0.00202	1,82 (*)	-0,00706	-1,56	0,01102	1,18	-0,00348	-0,26		
adjusted R <sup>2</sup>	0.52%		8.03%		0.27%		1.93%		0,49%		7,88%		0,26%		1.86%			

**Note:** This table shows the results of the estimation of Eq (8) using the GRI as explanatory variable disaggregated by country: France (Panel A), Germany (Panel B), Italy (Panel C) and Spain (Panel D). Dependent variable: TEM (total earnings management). Disaggregation of TEM: DA (discretionary accruals), DCFO (discretionary cash flow from operations), DEXP (discretionary operating expenses). Control variables: ROA: Return on Assets; BtM: Book-to-Market ratio; DEBT: Level of indebtedness measured by Debt-to-Book value ratio; LN: Log-asset; Dummy variables: SEO: 1 when the company has increased capital in t, t+1 and t+2; otherwise, 0; GRI: 1 if the company complies with GRI standards and submits the information to external verification; 0 if it complies with GRI standards; FRQ\_GRI: Multiplicative variable that takes the value of 1 if the company has increased the capital and has the highest rating of GRI; takes a value of -1 if the company has increased the capital but has the lowest rating of GRI, and otherwise takes 0 value. (\*), (\*\*) and (\*\*\*) indicate that the parameter is statistically significant at a confidence level of 10%, 5% and 1%, respectively. Adjusted R2 indicates that the explanatory power of the model. GLS and OLS estimation results are provided.

who measure the degree of CSR of companies with the MERCO (Spanish Monitor of Corporate Reputation) index, which bases part of its evaluation on companies' monitoring of the GRI [114].

When we consider TEM, in French companies this variable increases with firm size and growth options. Furthermore, we found a positive relationship between TEM and CSR companies as measured by the GRI indicator, and a negative one for companies raising capital. For Italian firms, TEM increases with company size and decreases when RoA and growth options are positive, and in all countries there is an inverse relationship between TEM and profitability. The positive association between size and TEM is in line with the result obtained by Ref. [65]. The results for French companies show that the larger the growth opportunities the higher the propensity to practice TEM in line with [115].

As a consequence of the above, at a global level (TEM) only French companies show a statistically significant relationship between equity issuance and EM. In this case, moreover, the sign is negative because the component (DCFO) on which the manipulation is performed shows this sign when integrated in the total (see Eq. (1)).

# 6. Conclusion

Empirical evidence shows that managers engage in earnings management practices prior to capital increases. It has also been analysed whether socially responsible companies resort to EM practices, finding that those companies that are more committed to CSR are less involved in such practices. The objective of our study is to test whether socially responsible companies engage in EM practices in the periods prior to capital increases.

For this purpose, we analyse a sample of non-financial companies listed in France, Germany, Italy and Spain between 2012 and 2020. These are EU countries that use the same currency, thus avoiding exchange rate inconsistencies, and are subject to a common basic accounting framework. We incorporate a total of 12822 firms, with France having the largest number of firms, followed by Germany, Italy and Spain.

When estimating the explanatory models for total accruals, operating cash flows and operating expenses, we find that they all are statistically significant and that the parameters obtained for the explanatory variables differ across countries, which justifies our country-by-country analysis. However, for total accruals, in all countries we find a positive relationship with profitability and a negative association with depreciable assets. Regarding with operating cash flows, the association is positive with the increase in sales and sales for the fiscal year. Finally, in the operating expenses model, the two regressors show positive and statistically significant values for all countries.

To test our hypothesis we consider capital increases in the year or in any of the following two years, and we test the degree of the company's commitment to CSR in that period. In order to provide further robustness to the study, we have considered a double measure of CSR. We check the company's inclusion in the DJSI index, we also verify whether or not they prepare their sustainability reports in accordance with the GRI criteria, and whether or not they have been verified by an external agent. Furthermore, we have not only presented the results using total earnings management (TEM), but also its decomposition into its three components (discretionary accruals -DA-, abnormal cash-flows from operations -DCFO- and abnormal operating expenses -DEXP-).

We find that when the GRI variable is used as a measure of CSR quality, the goodness of fit of the model is slightly higher than when the DJSI variable is used, and that there is an inverse relationship between the size of the financial market and the explanatory power of the model.

Results show that, in Spanish companies, there is no relationship between TEM or any of its components with the capital increase in the following years, nor with CSR as measured by the two indeces used (DJSI and GRI). In the rest of the countries, in relation to the SEO variable we only find a positive relationship in the DCFO component, which means that companies in France, Germany and Italy increase their DCFO in the years prior to the capital increase. These results were foreseeable according to previous studies; however, it is necessary to go deeper into the analysis of Spanish companies, where the results obtained have not been significant.

If we focus on the CSR level variable, only French companies show a decrease in DCFO as the quality and development of their CSR increases, according to the two indices considered. As a result, when analysing the TEM and capital increases, we observe that there is only a significant and negative relationship for French companies.

Our research provides a first contribution to the understanding of the behaviour of socially responsible companies with regard to performance management in an SEO environment, although it is subject to certain limitations. Borralho et al. [71] show that ESG components (environmental, social and corporate governance) have different impact on the EM. Therefore, our results can be contrasted by applying these criteria as measures of corporate social commitment. Another limitation is the number of countries analysed, as well as not considering the sector of activity of the companies, aspects that can be studied in future research. Although all countries studied share a common legal framework and a common currency, the results obtained differ between countries. Future research can include specific factors to identify the causes of these behaviours.

Despite the limitations, this study represents a breakthrough in the analysis of EM, as its relationship with CSR and SEO has so far been studied separately. However, it is interesting for academics, analysts and investors to know whether companies considered socially responsible engage in earnings management practices before going to the capital market to obtain financing through SEO operations. This study could be extrapolated, with the necessary adaptations, to any other environment or market.

## Author contribution statement

Mariano González-Sánchez; Ana Segovia-San Juan; Eva M. Ibáñez Jiménez: Conceived and designed the experiments; Performed the experiments; Analysed and interpreted the data; Contributed reagents, materials, analysis tools or data; Wrote the paper.

#### Data availability statement

Data included in article/supp. material/referenced in article.

## Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

#### References

- [1] K. Schipper, Earnings management, account, Horiz 3 (1989) 91–102.
- [2] P.M. Dechow, R.G. Sloan, A.P. Sweeney, Detecting Earnings Management, Account, . Rev., 1995, pp. 193-225.
- [3] K.A. Gunny, The relation between earnings management using real activities manipulation and future performance: evidence from meeting earnings benchmarks, Contemp. Account. Res. 27 (2010) 855–888, https://doi.org/10.1111/j.1911-3846.2010.01029.x.
- [4] S. Roychowdhury, Earnings management through real activities manipulation, J. Account. Econ. 42 (2006) 335–370, https://doi.org/10.1016/j.
- jacceco.2006.01.002. [5] S.P. Kothari, A.J. Leone, C.E. Wasley, Performance matched discretionary accrual measures, J. Account. Econ. 39 (2005) 163–197, https://doi.org/10.1016/j. jacceco.2004.11.002.
- [6] B.C. Ayers, J. Jiang, P.E. Yeung, Discretionary accruals and earnings management: an analysis of pseudo earnings targets, Account. Rev. 81 (2006) 617–652, https://doi.org/10.2308/accr.2006.81.3.617.
- [7] P.M. Dechow, A.P. Hutton, J.H. Kim, R.G. Sloan, Detecting earnings management: a new approach, J. Account. Res. 50 (2012) 275–334, https://doi.org/ 10.1111/j.1475-679X.2012.00449.x.
- [8] P.M. Healy, J.M. Wahlen, A review of the earnings management literature and its implications for standard setting, Account. Horiz. 13 (1999) 365–383, https://doi.org/10.2308/acch.1999.13.4.365.
- [9] W.N. Davidson III, P. Jiraporn, Y.S. Kim, C. Nemec, Earnings management following duality-creating successions: ethnostatistics, impression management, and agency theory, Acad. Manag. J. 47 (2004) 267–275, https://doi.org/10.5465/20159577.
- [10] P. Jiraporn, G.A. Miller, S.S. Yoon, Y.S. Kim, Is earnings management opportunistic or beneficial? An agency theory perspective, Int. Rev. Financ. Anal. 17 (2008), https://doi.org/10.1016/j.irfa.2006.10.005.
- [11] Y. Man, Earnings management and agency costs: is China different? J. Corp. Account. Finance 32 (2021) 13–30, https://doi.org/10.1002/jcaf.22481.
- [12] M.L. DeFond, C.W. Park, Smoothing income in anticipation of future earnings, J. Account. Econ. 23 (1997) 115–139, https://doi.org/10.1016/S0165-4101 (97)00004-9.
- [13] R. Kasznik, On the association between voluntary disclosure and earnings management, J. Account. Res. 37 (1999) 57–81, https://doi.org/10.2307/2491396.
- [14] M.S. Park, T. Park, Insider sales and earnings management, J. Account. Publ. Pol. 23 (2004) 381-411, https://doi.org/10.1016/j.jaccpubpol.2004.07.003.
- [15] A. Beyer, Financial analysts' forecast revisions and managers' reporting behavior, J. Account. Econ. 46 (2008) 334–348, https://doi.org/10.1016/j. jacceco.2008.08.003.
- [16] G. Gong, L.Y. Li, J.J. Wang, Serial correlation in management earnings forecast errors, J. Account. Res. 49 (2011) 677–720, https://doi.org/10.1111/j.1475-679X.2011.00407.x.
- [17] J.B. Kim, Accounting flexibility and managers' forecast behavior prior to seasoned equity offerings, Rev. Account. Stud. 21 (2016) 1361–1400, https://doi. org/10.1007/s11142-016-9372-0.
- [18] J. Sawicki, K. Shrestha, Insider trading and earnings management, J. Bus. Finance Account. 35 (2008) 331–346, https://doi.org/10.1111/j.1468-5957 2008 02075 x
- [19] X. Zhu, S. Lu, Earnings Management through Real Activities Manipulation before Mergers and Acquisitions, (n.d.).
- [20] C. Mellado-Cid, S.R. Jory, T. Ngo, Real earnings management activities prior to bond issuance, BRQ Bus. Res. Q. 20 (2017) 164–177, https://doi.org/10.1016/ j.brq.2017.04.004.
- [21] D. Agustia, N.P.A. Muhammad, Y. Permatasari, Earnings management, business strategy, and bankruptcy risk: evidence from Indonesia, Heliyon 6 (2020), e03317, https://doi.org/10.1016/j.heliyon.2020.e03317.
- [22] J.R. Ritter, The long-run performance of initial public offerings, J. Finance 46 (1991) 3–27, https://doi.org/10.1111/j.1540-6261.1991.tb03743.x.
- [23] D.K. Spiess, J. Affleck-Graves, Underperformance in long-run stock returns following seasoned equity offerings, J. Financ. Econ. 38 (1995) 243–267, https:// doi.org/10.1016/0304-405X(94)00817-K.
- [24] T. Loughran, J.R. Ritter, The operating performance of firms conducting seasoned equity offerings, J. Finance 52 (1997) 1823–1850, https://doi.org/10.2307/ 2329466.
- [25] S. Rangan, Earnings management and the performance of seasoned equity offerings, J. Financ. Econ. 50 (1998) 101–122, https://doi.org/10.1016/S0304-405X(98)00033-6.
- [26] S.H. Teoh, I. Welch, T.J. Wong, Earnings management and the underperformance of seasoned equity offerings, J. Financ. Econ. 50 (1998) 63–99, https://doi. org/10.1016/S0304-405X(98)00032-4.
- [27] S.H. Teoh, I. Welch, T.J. Wong, Earnings management and the long-run market performance of initial public offerings, J. Finance 53 (1998) 1935–1974, https://doi.org/10.1111/0022-1082.00079.
- [28] S.P. Kothari, N. Mizik, S. Roychowdhury, Managing for the moment: the role of earnings management via real activities versus accruals in SEO valuation, Account. Rev. 91 (2016) 559–586, https://doi.org/10.2308/accr-51153.
- [29] T.-H. Yang, J. Hsu, W.-B. Yang, Firm's motives behind SEOs, earnings management, and performance, Int. Rev. Econ. Finance 43 (2016) 160–169, https://doi. org/10.1016/j.iref.2015.10.038.
- [30] G. Wang, M. Hagigi, The effect of the need for subsequent seasoned equity offerings on earnings management motivation, Rev. Account. Finance 18 (2019) 25–52, https://doi.org/10.1108/RAF-01-2018-0019.
- [31] A. Duc Ngo, O. Varela, Earnings smoothing and the underpricing of seasoned equity offerings, Manag. Finance 38 (2012) 833–859, https://doi.org/10.1108/ 03074351211248180.
- [32] P.-G. Shu, S.-J. Chiang, Firm size, timing, and earnings management of seasoned equity offerings, Int. Rev. Econ. Finance 29 (2014) 177–194, https://doi.org/ 10.1016/j.iref.2013.05.011.
- [33] K. Yuan, D. Zeng, X. Yuan, F. Lan, Real earnings management, manipulation incentives and accounting conservatism: evidence from China, Emerg. Mark. Finance Trade 58 (2022) 939–951, https://doi.org/10.1080/1540496X.2020.1852927.
- [34] D.A. Cohen, P. Zarowin, Accrual-based and real earnings management activities around seasoned equity offerings, J. Account. Econ. 50 (2010) 2–19, https:// doi.org/10.1016/j.jacceco.2010.01.002.
- [35] C.-H. Chang, H.-W.W. Lin, Does there prevail momentum in earnings management for seasoned equity offering firms? Int. Rev. Econ. Finance 55 (2018) 111–129, https://doi.org/10.1016/j.iref.2017.12.015.
- [36] J. Martínez-Ferrero, S. Banerjee, I.M. García-Sánchez, Corporate social responsibility as a strategic shield against costs of earnings management practices, J. Bus. Ethics JBE. 133 (2016) 305–324, https://doi.org/10.1007/s10551-014-2399-x.

- [37] J. Martínez-Ferrero, D. Ruiz-Cano, I.-M. García-Sánchez, The causal link between sustainable disclosure and information asymmetry: the moderating role of the stakeholder protection context, Corp. Soc. Responsib. Environ. Manag. 23 (2016) 319–332, https://doi.org/10.1002/csr.1379.
- [38] J.F. Egginton, G.A. McBrayer, Does it pay to be forthcoming? Evidence from CSR disclosure and equity market liquidity, Corp. Soc. Responsib. Environ. Manag. 26 (2019) 396–407, https://doi.org/10.1002/csr.1691.
- [39] Z. Li, P. Wang, Flotation costs of seasoned equity offerings: does corporate social responsibility matter? Eur. Financ. Manag. 28 (2022) 771–808, https://doi. org/10.1111/eufm.12327.
- [40] Y. Kim, S. Li, C. Pan, L. Zuo, The role of accounting conservatism in the equity market: evidence from seasoned equity offerings, account, Rev 88 (2013) 1327–1356.
- [41] Z.-Y. Feng, C.R. Chen, Y.-J. Tseng, Do capital markets value corporate social responsibility? Evidence from seasoned equity offerings, J. Bank. Finance 94 (2018) 54–74, https://doi.org/10.1016/j.jbankfin.2018.06.015.
- [42] Y. Hong, M.L. Andersen, The relationship between corporate social responsibility and earnings management: an exploratory study, J. Bus. Ethics 104 (2011) 461–471.
- [43] B. Scholtens, F.-C. Kang, Corporate social responsibility and earnings management: evidence from Asian economies, Corp. Soc. Responsib. Environ. Manag. 20 (2013) 95–112, https://doi.org/10.1002/csr.1286.
- [44] S. Honggowati E. Supriyono Rahmawati, The effect of corporate social responsibility on financial performance with real manipulation as a moderating variable, Int. J. Manag. Econ. Soc. Sci. IJMESS. 3 (2014) 59–78.
- [45] E. Gras-Gil, M.P. Manzano, J.H. Fernández, Investigating the relationship between corporate social responsibility and earnings management: evidence from Spain, BRQ Bus. Res. Q. 19 (2016) 289–299, https://doi.org/10.1016/j.brq.2016.02.002.
- [46] T.D. Fields, T.Z. Lys, L. Vincent, Empirical research on accounting choice, J. Account. Econ. 31 (2001) 255–307, https://doi.org/10.1016/S0165-4101(01) 00028-3.
- [47] B. Al-Shattarat, K. Hussainey, W. Al-Shattarat, The impact of abnormal real earnings management to meet earnings benchmarks on future operating performance, Int. Rev. Financ. Anal. 81 (2022), 101264, https://doi.org/10.1016/j.irfa.2018.10.001.
- [48] Y. Zhao, K.H. Chen, Y. Zhang, M. Davis, Takeover protection and managerial myopia: evidence from real earnings management, J. Account. Publ. Pol. 31 (2012) 109–135, https://doi.org/10.1016/j.jaccpubpol.2011.08.004.
- [49] A. Zang, Evidence on the Trade-Off between Real Activities Manipulation and Accrual-Based Earnings Management, 2011. https://papers.ssrn.com/ abstract=1891759. (Accessed 28 January 2023).
- [50] J.R. Graham, C.R. Harvey, S. Rajgopal, The economic implications of corporate financial reporting, J. Account. Econ. 40 (2005) 3–73, https://doi.org/ 10.1016/j.jacceco.2005.01.002.
- [51] J.J. Griffin, J.F. Mahon, The corporate social performance and corporate financial performance debate: twenty-five years of incomparable research, Bus. Soc. 36 (1997) 5–31, https://doi.org/10.1177/000765039703600102.
- [52] M. Palacios-Mazano, E. Gras-Gil, J.M. Santos-Jaen, Corporate social responsibility and its effect on earnings management: an empirical research on Spanish firms, Total Qual. Manag. Bus. Excel. 32 (2021) 921–937, https://doi.org/10.1080/14783363.2019.1652586.
- [53] G. Heal, Corporate social responsibility: an economic and financial framework, Geneva Pap, Risk Insur.-Issues Pract. 30 (2005) 387–409, https://doi.org/ 10.1057/palgrave.gpp.2510037.
- [54] M.A. Harjoto, H. Jo, Corporate governance and CSR nexus, J. Bus. Ethics 100 (2011) 45–67, https://doi.org/10.1007/s10551-011-0772-6.
- [55] R.C. Chen, S.-W. Hung, C.-H. Lee, Does corporate value affect the relationship between Corporate Social Responsibility and stock returns? J. Sustain. Finance Invest. 7 (2017) 188–196, https://doi.org/10.1080/20430795.2016.1272947.
- [56] K. Lopatta, F. Canitz, S.A. Tideman, Abnormal CSR and financial performance, Eur. Account. Rev. 0 (2022) 1–27, https://doi.org/10.1080/ 09638180.2022.2084134.
- [57] J.M. Ford, S.A. Gehricke, J.E. Zhang, Option traders are concerned about climate risks: ESG ratings and short-term sentiment, J. Behav. Exp. Finance. 35 (2022), 100687, https://doi.org/10.1016/j.jbef.2022.100687.
- [58] G. Friede, T. Busch, A. Bassen, ESG and financial performance: aggregated evidence from more than 2000 empirical studies, J. Sustain. Finance Invest. 5 (2015) 210–233, https://doi.org/10.1080/20430795.2015.1118917.
- [59] H.-L. Chih, C.-H. Shen, F.-C. Kang, Corporate social responsibility, investor protection, and earnings management: some international evidence, J. Bus. Ethics 79 (2008) 179–198, https://doi.org/10.1007/s10551-007-9383-7.
- [60] A.B. Amar, S. Chakroun, Do dimensions of corporate social responsibility affect earnings management? Evidence from France, J. Financ. Report. Account. 16 (2018) 348–370, https://doi.org/10.1108/JFRA-05-2017-0033.
- [61] S. Chakroun, A.B. Amar, Earnings management, financial performance and the moderating effect of corporate social responsibility: evidence from France, Manag. Res. Rev. 45 (2021) 331–362, https://doi.org/10.1108/MRR-02-2021-0126.
- [62] S. Bozzolan, M. Fabrizi, C.A. Mallin, G. Michelon, Corporate social responsibility and earnings quality: international evidence, Int. J. Account. 50 (2015) 361–396, https://doi.org/10.1016/j.intacc.2015.10.003.
- [63] Y. Almahrog, Z. Ali Aribi, T. Arun, Earnings management and corporate social responsibility: UK evidence, J. Financ. Report. Account. 16 (2018) 311–332, https://doi.org/10.1108/JFRA-11-2016-0092.
- [64] L. Moratis, M. van Egmond, Concealing social responsibility? Investigating the relationship between CSR, earnings management and the effect of industry through quantitative analysis, Int. J. Corp. Soc. Responsib. 3 (2018) 8, https://doi.org/10.1186/s40991-018-0030-7.
- [65] P.E. Dimitropoulos, Corporate social responsibility and earnings management in the EU: a panel data analysis approach, Soc. Responsib. J. 18 (2020) 68–84, https://doi.org/10.1108/SRJ-04-2020-0156.
- [66] R.C. Chen, S.-W. Hung, Exploring the impact of corporate social responsibility on real earning management and discretionary accruals, Corp. Soc. Responsib. Environ. Manag. 28 (2021) 333–351, https://doi.org/10.1002/csr.2052.
- [67] P. Velte, The bidirectional relationship between ESG performance and earnings management empirical evidence from Germany, J. Glob. Responsib. 10 (2019) 322–338, https://doi.org/10.1108/JGR-01-2019-0001.
- [68] Z. Khelil-Rhouma, M. Hamed-Sidhom, Earnings management and the quality of non-financial reporting in a regulated context, Corp. Gov. Sustain. Rev. 5 (2021) 56, https://doi.org/10.22495/cgsrv5i4p5.
- [69] Y. Chouaibi, G. Zouari, The mediating role of real earnings management in the relationship between CSR practices and cost of equity: evidence from European ESG data, EuroMed J. Bus. (2022), https://doi.org/10.1108/EMJB-12-2021-0183 ahead-of-print.
- [70] R. Pathak, R. Das Gupta, Environmental, social and governance performance and earnings management the moderating role of law code and creditor's rights, Finance Res. Lett. 47 (2022), 102849, https://doi.org/10.1016/j.frl.2022.102849.
- [71] J.M. Borralho, R. Hernández-Linares, D. Gallardo-Vázquez, I. Choban de Sousa Paiva, Environmental, social and governance disclosure's impacts on earnings management: family versus non-family firms, J. Clean. Prod. 379 (2022), 134603, https://doi.org/10.1016/j.jclepro.2022.134603.
- [72] M. Nagy, K. Valaskova, P. Durana, The effect of CSR policy on earnings management behavior: evidence from visegrad publicly listed enterprises, Risks 10 (2022) 203, https://doi.org/10.3390/risks10110203.
- [73] M. Khenissi, A. Jahmane, M. Hofaidhllaoui, Does the introduction of CSR criteria into CEO incentive pay reduce their earnings management? The case of companies listed in the SBF 120, Finance Res. Lett. 48 (2022), 102880, https://doi.org/10.1016/j.frl.2022.102880.
- [74] S. Khlifi, G. Zouari, Real Earnings Management, CSR and the Moderating Effect of Corporate Governance in Innovative Firms, vol. 5, 2021.
- [75] S. Khlifi, G. Zouari, The moderating role of good corporate governance on the relationship between corporate social responsibility and real earnings management, J. Account. Manag. Inf. Syst. 21 (2022) 524–545, https://doi.org/10.24818/jamis.2022.04003.
- [76] M. Toukabri, I. Kateb, The bidirectional relationship between corporate social responsibility and financial reporting quality: Does gender diversity matter? Evidence from US companies, J. Corp. Account. Finance. n/a (n.d.). https://doi.org/10.1002/jcaf.22603.

- [77] D. Prior, J. Surroca, J.A. Tribó, Are socially responsible managers really ethical? Exploring the relationship between earnings management and corporate social responsibility, Corp. Gov. Int. Rev. 16 (2008) 160–177, https://doi.org/10.1111/j.1467-8683.2008.00678.x.
- [78] V. Grougiou, S. Leventis, E. Dedoulis, S. Owusu-Ansah, Corporate social responsibility and earnings management in US banks, in: Account. Forum, Elsevier, 2014, pp. 155–169, https://doi.org/10.1016/j.accfor.2014.05.003.
- [79] E. López-González, J. Martínez-Ferrero, E. García-Meca, Does corporate social responsibility affect earnings management? Evidence from family firms, Rev. Contab. Span. Account. Rev. RC-SAR. 22 (2019) 233–247, https://doi.org/10.6018/rcsar.349601.
- [80] L. Fauver, G. Loureiro, A.G. Taboada, The impact of regulation on information quality and performance around seasoned equity offerings: international evidence, J. Corp. Finance 44 (2017) 73–98, https://doi.org/10.1016/j.jcorpfin.2017.03.008.
- [81] N.A. Preussner, E. Aschauer, The accuracy and informativeness of management earnings forecasts: a review and unifying framework, Account. Perspect. 21 (2022) 273–330, https://doi.org/10.1111/1911-3838.12294.
- [82] J. Bertomeu, E. Cheynel, E.X. Li, Y. Liang, How pervasive is earnings management? Evidence from a structural model, Manag. Sci. 67 (2021) 5145–5162, https://doi.org/10.1287/mnsc.2020.3717.
- [83] S. Opare, M. Houqe, T. van Zijl, Earnings management and underperformance after seasoned equity offerings: a cross-country study, Int. J. Manag. Finance (2022), https://doi.org/10.1108/IJMF-02-2022-0075 ahead-of-print.
- [84] R.-S. Wu, H.-C. Lo, A Rose has its Thorn: the role of institutional investors in SEO firms' reporting behavior, Int. Rev. Econ. Finance 82 (2022) 530-554, https://doi.org/10.1016/j.iref.2022.07.004.
- [85] P.T. Chan, T. Walter, Investment performance of "environmentally-friendly" firms and their initial public offers and seasoned equity offers, J. Bank. Finance 44 (2014) 177–188, https://doi.org/10.1016/j.jbankfin.2014.04.006.
- [86] A.J. Hillman, G.D. Keim, Shareholder value, stakeholder management, and social issues: what's the bottom line? Strateg. Manag. J. 22 (2001) 125–139.
- [87] J.E. Mattingly, Corporate social performance: a review of empirical research examining the corporation-society relationship using kinder, lydenberg, domini social ratings data, Bus. Soc. 56 (2017) 796–839, https://doi.org/10.1177/0007650315585761.
- [88] M. Nizamuddin, Corporate social responsibility and corporate financial performance: an exploratory study of measurement-approach selection issues, IUP J. Corp. Gov. 17 (2018) 36–54.
- [89] H. Adıgüzel, D. Özbay, Corporate reputation and real activities management: evidence from an emerging economy, J Bus Econ Pol 4 (2017) 77-89.
- [90] M.V. López, A. Garcia, L. Rodriguez, Sustainable development and corporate performance: a study based on the Dow Jones sustainability index, J. Bus. Ethics JBE. 75 (2007) 285–300, https://doi.org/10.1007/s10551-006-9253-8.
- [91] H.-L. Chih, H.-H. Chih, T.-Y. Chen, On the determinants of corporate social responsibility: international evidence on the financial industry, J. Bus. Ethics 93 (2010) 115–135, https://doi.org/10.1007/s10551-009-0186-x.
- [92] I.C. Lourenço, M.C. Branco, J.D. Curto, T. Eugénio, How does the market value corporate sustainability performance? J. Bus. Ethics JBE. 108 (2012) 417–428, https://doi.org/10.1007/s10551-011-1102-8.
- [93] B.S. Magnanelli, M.F. Izzo, Corporate social performance and cost of debt: the relationship, Soc. Responsib. J. 13 (2017) 250–265, https://doi.org/10.1108/ SRJ-06-2016-0103.
- [94] I.D. del Valle, J.M. Diez Esteban, O.L. de Foronda Perez, Corporate social responsibility and sustainability committee inside the board, Eur. J. Int. Manag. 13 (2019) 159–176, https://doi.org/10.1504/EJIM.2019.098145.
- [95] S. Lee, I. Kim, C. Hong, Who values corporate social responsibility in the Korean stock market? Sustainability 11 (2019) 5924, https://doi.org/10.3390/ sul1215924.
- [96] A. Willis, The role of the global reporting initiative's sustainability reporting guidelines in the social screening of investments, J. Bus. Ethics 43 (2003) 233–237, https://doi.org/10.1023/A:1022958618391.
- [97] P.M. Clarkson, Y. Li, G.D. Richardson, F.P. Vasvari, Revisiting the relation between environmental performance and environmental disclosure: an empirical analysis, Account, Organ. Soc. 33 (2008) 303–327, https://doi.org/10.1016/j.aos.2007.05.003.
- [98] H. Schadewitz, M. Niskala, Communication via responsibility reporting and its effect on firm value in Finland, Corp. Soc. Responsib. Environ. Manag. 17 (2010) 96–106, https://doi.org/10.1002/csr.234.
- [99] J.L. Miralles Marcelo, M. del Mar Miralles Quirós, J.L. Miralles Quirós, Performance bursátil de las empresas socialmente responsables, Cuad. Econ. Dir. Empresa CEDE. 15 (2012) 221–230, https://doi.org/10.1016/j.cede.2012.06.002.
- [100] M. Ali, K. Hameedi, A. Almagtome, Does sustainability reporting via accounting information system influence the investment decisions in Iraq, Int. J. Innov. Creat. Change. 9 (2019) 294–312.
- [101] A. Ali, L.-S. Hwang, Country-specific factors related to financial reporting and the value relevance of accounting data, J. Account. Res. 38 (2000) 1–21, https:// doi.org/10.2307/2672920.
- [102] E. Bartov, S.R. Goldberg, M.-S. Kim, The valuation-relevance of earnings and cash flows: an international perspective, J. Int. Financ. Manag. Account. 12 (2001) 103–132, https://doi.org/10.1111/1467-646X.00068.
- [103] P. Delvaille, G. Ebbers, C. Saccon, International financial reporting convergence: evidence from three continental European countries, account, Eur 2 (2005) 137–164, https://doi.org/10.1080/09638180500379103.
- [104] A. Devalle, E. Onali, R. Magarini, Assessing the value relevance of accounting data after the introduction of IFRS in Europe, J. Int. Financ. Manag. Account. 21 (2010) 85–119, https://doi.org/10.1111/j.1467-646X.2010.01037.x.
- [105] M. Troilo, B.R. Walkup, M. Abe, S. Lee, Legal systems and the financing of working capital, Int. Rev. Econ. Finance 64 (2019) 641–656, https://doi.org/ 10.1016/j.iref.2018.01.010.
- [106] J.Q. Li, Y. Ding, Institutional effects on information content of US and French management earnings forecasts: evidence from market reactions and analyst revisions, Adv. Account. 24 (2008) 101–109, https://doi.org/10.1016/j.adiac.2008.05.004.
- [107] M. Arellano, S. Bond, Some tests of specification for panel data: Monte Carlo evidence and an application to employment equations, Rev. Econ. Stud. 58 (1991) 277–297, https://doi.org/10.2307/2297968.
- [108] E.F. Fama, K.R. French, The cross-section of expected stock returns, J. Finance 47 (1992) 427-465, https://doi.org/10.2307/2329112.
- [109] J.M. Griffin, M.L. Lemmon, Book-to-Market equity, distress risk, and stock returns, J. Finance 57 (2002) 2317–2336, https://doi.org/10.1111/1540-6261.00497.
- [110] M. Vassalou, Y. Xing, Default risk in equity returns, J. Finance 59 (2004) 831-868, https://doi.org/10.1111/j.1540-6261.2004.00650.x.
- [111] M.F. McNichols, Research design issues in earnings management studies, J. Account. Publ. Pol. 19 (2000) 313–345, https://doi.org/10.1016/S0278-4254(00) 00018-1.
- [112] C.M. Cupertino, A.L. Martinez, N.C.A. da Costa, Earnings manipulations by real activities management and investors' perceptions, Res. Int. Bus. Finance 34 (2015) 309–323, https://doi.org/10.1016/j.ribaf.2015.02.015.
- [113] H. Halabi, A. Alshehabi, I. Zakaria, Informal institutions and managers' earnings management choices: evidence from IFRS-adopting countries, J. Contemp. Account. Econ. 15 (2019), 100162, https://doi.org/10.1016/j.jcae.2019.100162.
- [114] El proceso de elaboración de Merco España, 2013. Monitor empresarial de reputación corporativa. https://www.merco.info/files/2015/10/106/metodologiae-informe-de-verificacion-merco-talento-es-2013.pdf, 2013.
- [115] T. Gonçalves, C. Gaio, A. Ferro, Corporate social responsibility and earnings management: moderating impact of economic cycles and financial performance, Sustainability 13 (2021) 9969, https://doi.org/10.3390/su13179969.